QUANTIFYING AGE-RELATED BONE LOSS AND VITAMIN D DEFICIENCY

QUANTIFYING AGE-RELATED BONE LOSS AND VITAMIN D DEFICIENCY: EXPLORING CO-OCCURRENCE AND CO-MORBIDITY IN TWO 18TH-19TH CENTURY FRENCH-CANADIAN COMMUNITIES

BY REBECCA CHRISTENSON, B.A. (HONS)

A Thesis Submitted to the School of Graduate Studies in Partial Fulfilment of the Requirements

for the Degree Master of Arts

McMaster University © Copyright by Rebecca Christenson, September 2023

McMaster University

MASTER OF ARTS (2023)

Hamilton, Ontario, Canada (Anthropology)

TITLE: Quantifying Age-Related Bone Loss and Vitamin D Deficiency: Exploring Co-Occurrence and Co-Morbidity in Two 18th-19th Century French-Canadian Communities

AUTHOR: Rebecca Christenson

B.A. (Hons) (University of Alberta)

SUPERVISOR: Dr. Megan Brickley

NUMBER OF PAGES: xii, 173

Lay Abstract

This study examines skeletal health in two historical French-Canadian communities using digital X-ray images of the metacarpal bone of the index finger, and micro-CT scans of ribs. A sample of 43 individuals was chosen for X-ray analysis. Each individual's bone amount was calculated from the X-ray images, and further analysis revealed there were no significant differences in bone amount between males and females, and no differences between young adults, middle-aged adults, and older adults. Ten individuals were chosen from the sample of 43 to examine of bone quality using rib scans. Seven individuals showed signs of low-quality bone. Assessing the results alongside historical records suggests that social forces may have interacted with the biological systems that maintain skeletal health.

Abstract

This study analyzed age and sex-related bone quantity and quality patterns in the historical French-Canadian communities of Pointe-aux-Trembles and Sainte-Marie-de-Beauce. The analyses were undertaken using newly developed techniques in metacarpal radiogrammetry and rib micro-CT assessment. The patterns of bone quantity and quality were used to infer biological and social contributions to individual and community health in the past.

Second metacarpals from a sample of 43 individuals from Pointe-aux-Trembles (f=12, m=6) and Sainte-Marie (f=15, m=10) were radiographed and assessed. A 1.9cm region of interest (ROI) was isolated on the diaphyseal surface. Within this ROI, cortical and total bone areas were measured and used to calculate the cortical area index (CA_{ROI}). Patterns of bone amount by age category (15-29, 30-49, 50+) and sex were statistically analyzed. There were no statistically significant differences in CAI_{ROI} between males and females, and no significant differences were found between any of the age categories.

From the total sample of 43 individuals, rib fragments from a sub-sample of 10 individuals were chosen for micro-CT analysis. Five micro-CT slices along a 1cm segment of bone were examined and assigned scores for the appearance of mineralization defects associated with osteomalacia. Scores obtained were examined alongside the age, sex, and CAI_{ROI} data. Seven individuals in this sub-sample showed consistent evidence of mineralization defects throughout their ribs, although no patterns were noted between the presence and severity of defects and age, sex, or CAI_{ROI}.

These results highlight the bioculturally complex nature of individual and community health in the past. Pointe-aux-Trembles and Sainte-Marie both experienced frequent periods of starvation, the spread of infectious disease from urban centers, and a potential lack of vitamins necessary to develop good-quality bone in adequate amounts. Analyzing these results alongside historical context provided insight into the long-term impacts of stress events on individual and community health. This thesis is dedicated to my very special cats, Mystery, Lady Grey, Orange Juice, and Mocha,

for reminding me that the best things in life aren't things.

Acknowledgements

I would like to express my deepest appreciation to my thesis supervisor, Dr. Megan Brickley, and my committee member, Dr. Tina Moffat, for their guidance and support throughout not only the research process but also during my move from Edmonton to Hamilton. I am incredibly grateful to Dr. Amanda Wissler for being an enthusiastic third reader and providing me with valuable feedback, and to Dr. Isabelle Ribot at the University of Montréal for generously hosting me in her lab and granting me access to the skeletal collections included in this thesis.

Special thanks to the faculty and staff at McMaster University, the entire Department of Anthropology, and my graduate cohort for supporting and encouraging me during difficult times. Thanks should also go to my close friends from faraway places, Dr. Emily Hull, for first introducing me to human osteology, and Tyra Davidson, for always being willing to read my drafts. Their encouragement and insights have been invaluable to me.

Lastly, I would like to thank my parents, Debbie and Richard Christenson, for celebrating me and reminding me of my strength, and for teaching me that I can do anything, no matter how difficult it may be.

vi

Table	of	Contents
	•••	0011001100

LAY ABSTRACTiii
ABSTRACTiv
DEDICATIONv
ACKNOWLEDGEMENTSvi
LIST OF FIGURESix
LIST OF TABLESxi
DECLARATION OF ACADEMIC ACHIEVEMENTxii
CHAPTER 1: INTRODUCTION1
1.1 Introduction1
1.2 Thesis Organization6
CHAPTER 2: BACKGROUND7
2.1 Theoretical Orientation7
2.2 Social and Historical Context8
2.3 The Bioarchaeology of Osteoporosis10
2.4 Vitamin D Deficiency, Osteomalacia, and Osteoporosis
CHAPTER 3: MATERIALS AND METHODS22
3.1 Materials22
3.2 Methods26
CHAPTER 4: RESULTS
4.1 Introduction36
4.2 Metacarpal Radiogrammetry36

4.3 Micro-CT	Analysis	39
--------------	----------	----

CHAPTER 5: DISCUSSION47
5.1 Introduction47
5.2 Metacarpal Radiogrammetry47
5.3 Osteobiographical Accounts55
5.4 Osteoporosis, Vitamin D Deficiency, and Community Health in Quebec68
5.5 Limitations and Future Research72
CHAPTER 6: CONCLUSION74
REFERENCES77
APPENDICES

List of Figures

Chapter 3: Materials and Methods

Figure 3.1. Map of Montréal.	23
Figure 3.2. Map of Québec.	25
Figure 3.3. Appearance of defective cement lines.	34
Figure 3.4. Appearance of incomplete mineralization associated with defective cement lines.	34
Chapter 4: Results	
Figure 4.1. Box plot of cortical area index in relation to age.	38

Figure 4.2. Scatter plot depicting the relationship between age and cortical area index.	38
Figure 4.3. Defective cement lines (DCL) in individual 7A2-S2, slice 1300.	41
Figure 4.4. Defective cement lines (DCL) in individual 7A2-S20, slice 1350.	41
Figure 4.5. Defective cement lines (DCL) in individual 7A9-S46, slice 450.	43
Figure 4.6. Defective cement lines (DCL) in individual 2E13, slice 1300 (A) and 1800 (B).	45

Chapter 5: Discussion

Figure 5.1. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 7A2-S2 in the sub-sample.	56
Figure 5.2. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 7A2-S11 in the sub-sample.	57
Figure 5.3. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 7A2-S20 in the sub-sample.	59

Figure 5.4. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 7A9-S38 in the sub-sample.	60
Figure 5.5. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 7A9-S46 in the sub-sample.	61
Figure 5.6. Cortical area index CAI _{ROI} and mineralization defect score of Individual 7A11-S19 in the sub-sample.	63
Figure 5.7. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 7A11-S42 in the sub-sample.	64
Figure 5.8. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 2B9 in the sub-sample.	65
Figure 5.9. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 2E13 in the sub-sample.	66
Figure 5.10. Cortical area index (CAI _{ROI}) and mineralization defect score of Individual 2G4 in the sub-sample.	67

List of Tables

Chapter 2: Background

Table 2.1. T- and Z-score values for the diagnosis of osteoporosis.	16
0 1	

Chapter 3: Materials and Methods

Table 3.1. Dates of cemetery use at Pointe-aux-Trembles and Sainte-Marie.	24
Table 3.2. Sample distribution by sex.	26
Table 3.3. Sample distribution by age.	26
Table 3.4. Defect scores and descriptions for slice analysis.	33
Table 3.5. Defect category derived from total defect score.	35

Chapter 4: Results

Table 4.1. Summary of age, sex, cortical bone area (CAI _{ROI}), and mineralization	46
defect scores.	

Chapter 5: Discussion

Table 5.1. Summary of observed age and sex trends in comparative studies.	49
---	----

Declaration of Academic Achievement

The research featured in this thesis was completed by Rebecca Christenson under the supervision of Dr. Megan Brickley. Research questions and methodology were developed in consultation with Dr. Brickley and Dr. Tina Moffat. Permission to access the skeletal collections was obtained by Dr. Brickley from Dr. Isabelle Ribot at the University of Montréal. Ethics approval was granted by the Human Tissue Committee of the Hamilton Integrated Research Ethics Board (HiREB) as part of the project "Health across the Life Course: An Investigation of the Impact of the Maternal-Child Nexus and Lifeways on Non-Adult and Adult Health" (ID# 16436).

Chapter 1: Introduction

1.1 Introduction

Osteoporosis and osteomalacia are often clinically considered to be complications of aging; however, this is a modern perspective on bone health primarily from communities with developed healthcare systems. These metabolic bone disorders occur as a consequence of interruptions to the metabolic systems responsible for bone maintenance, such as the mechanisms that utilize vitamin D. Vitamin D is primarily synthesized in the skin, stimulated by UVB rays from the sun, and obtained from dietary sources where it is absorbed in the intestinal tract (Brickley et al. 2020, pp. 75-128; Minisola et al. 2021). Insufficient dietary or cutaneous intake of vitamin D or the inability to convert vitamin D to its active form impairs the process of bone mineralization (Minisola et al. 2021).

Osteoporosis is characterized by the weakening of skeletal structures through the "uncoupling" of bone resorption and formation during bone maintenance and remodelling (Compston et al. 2019; Prestwood et al. 2004). This causes bones to become fragile and increases the likelihood of fracturing. According to the World Health Organization (1998), osteoporotic bone falls at least 2.5 standard deviations below the average young adult bone mineral density in a population.

There is a wide body of clinical and biomedical data detailing the breakdown in cellular and molecular processes that regulate bone maintenance during aging (Cremers et al. 2010; Doran and Khosla 2004; Kostenuik 2013; Misra and Silverberg 2010; Parfitt 2003, 2013; Prestwood and Raisz 2004; Zebaze and Seeman 2010), but non-age-related causes of

osteoporosis, such as nutrient deficiencies, insufficient mechanical stimuli, and inflammation, are not as well studied or understood by modern epidemiological researchers (Frost 2003; Prestwood and Raisz 2004). In contrast to osteoporosis, where bone becomes brittle, incomplete mineralization in osteomalacia results in soft bone that can become deformed (Brickley et al. 2020, pp. 75-128). Like osteoporosis, osteomalacia is associated with old age; the uptake of vitamin D and the conversion to its usable state becomes less efficient with age, and individual or cultural habits may limit the ability of the elderly to obtain vitamin D through diet or exposure to the sun (Lips 2001; Mithal et al. 2009). Although the physiological causes differ, osteomalacia and osteoporosis pose a great risk for fracture, which is associated with morbidity and mortality, particularly in elderly populations (Arnold 1973; Bischoff-Ferrari et al. 2006; International Osteoporosis Foundation 2023; Minisola et al. 2021; Schnell et al. 2010; Zebaze and Seeman 2010).

Biomedical studies show predictable and consistent patterns of changing bone metabolism with age, and the sexually dimorphic ways these processes occur (Kousteni 2010; Laurent et al. 2014; Poole et al. 2004; Riggs et al. 1998; Wiren 2005). These patterns can be observed in archaeological samples (Curate et al. 2013; Ericksen 1976; Glencross and Agarwal 2011; Van Gerven et al. 1969), but they are not universal (Agarwal and Grynpas 2009; Agarwal 2012; Beauchesne and Agarwal 2014, 2017; Dewey et al. 1969; Ekenman et al. 1995; Holck 2007; Poulsen et al. 2001; Zaki et al. 2009). Current clinical studies focus on the biological, primarily genetic and age-related, factors that contribute to conditions such as osteoporosis and osteomalacia (Duncan and Brown 2008; Giguère and Rousseau 2000; Ralston and Crombrugghe 2006), but often do not consider the impact of physical and social environments on the

development of such disorders (Agarwal 2012; Riancho and Brennan-Olsen 2017; Shavers 2007). These extrinsic factors can include diet, environmental and economic barriers to obtaining vitamins and nutrients, tobacco and alcohol consumption, exercise, chronic stress, body size, and pregnancy (Agarwal and Stuart-Macadam 2003; Agarwal 2019; Choi et al. 2021; Gough Courtney et al. 2023). Modern Western populations are privileged to experience osteoporosis and osteomalacia as side effects of aging. Access to foods that have been fortified with nutrients, from infancy to adulthood, reduces the risk of developing bone disorders due to malnutrition, and access to medical care allows the treatment of disorders that would otherwise impact skeletal health. However, many modern countries do not enforce or have access to food fortifications. Communities that lack food fortifications show widespread development of skeletal complications from malnutrition and disease at every stage of life (Jones 2018; Priemel et al. 2010).

The communities of colonial French-Canada (1700-1850) are ideal for exploring the social and physical factors that contribute to health alongside the biological components. The French-Canadian colonists typically worked outdoors, where the latitude along the St. Lawrence River could provide seasonal access to sufficient sunlight for vitamin D synthesis (Chaplin and Jablonski 2013; Greer 1997; Holick 2007; Huotari and Hertzig 2008; Lips 2001; Mithal et al. 2009; Ouellet 1980; Uday and Högler 2019). Fish from the St. Lawrence River would have been a reliable source of dietary vitamin D, an essential source of vitamin D from April to October when there is insufficient UVB available to stimulate the cutaneous production of vitamin D (Chaplin and Jablonski 2013; Lu et al. 2007; Vigeant et al. 2017).

The colonies were not egalitarian, but social order was not as strict as it was in Europe, leading to fewer class disparities and creating more equalized health experiences between the poor and the rich (Geloso et al. 2023; Greer 1997). However, from the establishment of the first colonies, experiences in New France were tumultuous. Harsh winters, constant war, crop failure, and economic fluctuations consistently impacted health and well-being in the colonies (Boulet 2021; Bruckner et al. 2018; Greer 1997; Ouellet 1980). Communities often experienced infectious disease epidemics and starvation, leading to periods of poor health that had both biological and cultural origins (Bruckner et al. 2018; Ouellet 1980).

The French-Canadian settlements at Pointe-aux-Trembles and Sainte-Marie were agricultural communities that experienced epidemics, starvation, and malnutrition from the 1700s to the 1800s (Ethnoscop 2006, 2016). During preliminary analyses of these skeletons, poor health and stress indicators were noted, making these communities valuable for exploring the links between nutritional deficiencies and skeletal health. In this study, 43 individuals from Pointe-aux-Trembles (*n*=18) and Sainte-Marie (*n*=25), were assessed for patterns of age-related bone loss using metacarpal radiogrammetry. Ten individuals were subsequently chosen from the total sample of 43 to undergo further analysis for evidence of osteomalacia using micro-CT scans of rib fragments. It was hypothesized that adherence to clinical age and sex patterns in bone loss would be observed, as these trends have been observed in comparable bioarchaeological studies (ex. Glencross and Agarwal 2011). Furthermore, it was expected that evidence of osteomalacia would be associated with osteoporosis, as disrupted and incomplete bone formation would result in low amounts of bone.

Bioarchaeologists rarely encounter complete, undamaged skeletons, and must make inferences based on fragmentary remains. As explored by Brickley and Buckberry (2015), even the most incomplete and poorly preserved remains can provide a range of valuable information regarding health and disease in the past. However, limitations in analytical methods may exclude fragmentary remains; for example, traditional metacarpal radiogrammetry requires complete metacarpals for analysis. Excluding incomplete elements limits the amount of data that can be analyzed, therefore limiting our understanding of past health and disease. Expanding methods of assessment to optimize the use of fragmentary remains allows bioarchaeologists to expand the amount of information that can be collected from poorly preserved and incomplete skeletons.

The aim of this thesis, then, is to utilize new methods of assessing bone quantity (metacarpal radiogrammetry) and quality (micro-CT) in fragmentary remains to answer the following questions:

1) What is the pattern of age-related bone loss in past French-Canadian communities, and what can this tell us about past health in these communities?

2) Could vitamin D deficiency contribute to the development of osteoporosis in past French-Canadian communities, and what can this add to our understanding of the co-occurrence and co-morbidity of osteoporosis and metabolic disorders?

3) How can a biocultural approach identify the cultural and physiological factors that influence bone health on both the community level and in the lived experience of individuals in

Pointe-aux-Trembles and Sainte-Marie?

1.2 Thesis Organization

This thesis is organized into six chapters and supplemented by additional appendices. This chapter has provided an outline of the aims of this study and its importance. Chapter Two provides the necessary historical background of colonial French Canada and an overview of the biological systems involved in developing healthy and pathological bone. Chapter Three provides details on the materials, methodologies, and statistical analyses that are used in this research, and Chapter 4 reports the results of these methods and the statistical analyses. Chapter 5 discusses the observed patterns reported in the previous chapter, and provides interpretations for these patterns from a biocultural perspective. This thesis is concluded with Chapter 6, which provides a summary of key findings and recommendations for further research.

Chapter 2: Background

2.1 Theoretical Orientation

This thesis utilizes two main theoretical frameworks: the biocultural approach and lived experience phenomenology. Biocultural approaches rely on a holistic and nuanced understanding of the relationships between culture, the environment, society, and physiology to synthesize interpretations of health and disease (Goodman and Leatherman 1998; Zuckerman and Martin 2016).

Lived experience analysis narrows the scope of biocultural practice to the individual level by interpreting an individual's experience within society (Boylorn 2008; Ellis and Flaherty 1992; Finlay 2009; van Manen 1997). This approach places lived experiences in direct comparison with one another, facilitating a nuanced understanding of how lives are shaped by shared and differing social factors such as gender, race, and class (Dahlberg 2021; Dodgson 2023; Russell 2007; van Manen 1997). In paleopathology, this approach aims to define and explore the social determinants of health that influence one's susceptibility to illness and ability to recover (Fay 2006; Knudson and Stojanowski 2008; Stodder and Palkovich 2012). In order to infer individual experiences and make direct health comparisons, lived experience was explored through an osteobiographical approach.

Hosek and Robb (2019) list several core components of osteobiographical data that contribute to understandings of the entire life course until death. Limitations in the present study design make it impossible to conduct a full life course analysis for each individual, but several of the core components listed by Hosek and Robb do fit within the methodological

framework established throughout this chapter. The key layers of data used to formulate the osteobiographies of these individuals include contextual information (physical, historical, and cultural), demographic data (age and sex), health status at death (mineralization defects and cortical bone amount), and postmortem history (interment).

2.2 Social and Historical Context

Eastern Canada was first colonized by the French in 1534 by Jacques Cartier on an expedition to find a passage to China. Although Cartier's journey did not take him to China, it made the French aware of the St. Lawrence region's rich natural resources and economic potential (Le Moine 1876; Lucas 1901).

In the late 1500s and early 1600s, the first French colonists established trading forts and military outposts that grew to become the urban centers of New France (Le Moine 1876). In the 1660s and 1670s, the French monarchy began to subsidize immigration to the colonies, incentivizing French peasants and servants to become farmers in the St. Lawrence region (Dechêne 1992; Greer 1997; Lucas 1901; Ouellet 1980). Land along the St. Lawrence River was organized into manors, owned by landlords (*seigneurs*) (Geloso et al. 2023; Greer 1997; Lucas 1901; Ouellet 1980). The manors were divided into long narrow strips, that were rented by working-class French settlers, the *habitants*, in exchange for money, labor, or produce (Dechêne 1992). *Habitants* were responsible for clearing their rented land of forests and establishing a farm, and *seigneurs* were required to provide a mill and a church on their land (Dechêne 1992; Dumont et al. 1987; Greer 1997). Farmers learned to cultivate corn, beans, and squash from the

local Iroquois population, and grew vegetables and tobacco for personal use. Each farm typically owned cows, pigs, and hens, providing the family with milk, beef, pork, and eggs (Dechêne 1992; Greer 1997). Wheat, flour, oats, and hay were the main agricultural export, although some families sold eggs and produce in the urban markets (Ouellet 1980). Urban centers were populated by the clergy and the colonial military, but the majority of the urban population consisted of artisans, beggars, day laborers, and vendors (Dechêne 1992; Dumont et al. 1987; Greer 1997). During prosperous periods, the people of New France could be relatively selfsufficient and free from the rigid social castes that existed in Europe at the time. The freedom offered by the French colonies was the main attraction for many settlers (Dumont et al. 1987).

Although life in the colonies may have appeared peaceful, the people of New France experienced many hardships. The most consistent challenge that the colonizers of French-Canada faced was adapting to the cold Canadian climate (Dechêne 1992; Ouellet 1980). Obtaining adequate nutrition in the winter was difficult without crops, and starving families occasionally resorted to killing their livestock animals for food. During the spring and summer, periods of crop failure and low yield occurred commonly throughout the communities of French Canada. Soil exhaustion, plant epidemics, and unfavorable weather conditions often impacted the harvest, leading to mass starvation in the colonies (Dechêne 1992; Dumont et al. 1987; Ouellet 1980). During times of conflict with the local Indigenous communities and colonial powers, fields were frequently destroyed, impacting the health and economy of the *habitants* occupying those areas (Brandão 1997; Greer 1997; Ouellet 1980). British naval blockades in the gulf of the St. Lawrence River during times of war effectively starved French armies by preventing the import of provisions from France, starving the *habitants* as well, as they had to

surrender their produce to the French armies, occasionally at gunpoint (Greer 1997; Marston 2001). Poor sanitation and hygiene resulted in frequent smallpox, measles, and cholera epidemics that began in urban centers and spread to rural communities (Dechêne 1992; Dumont et al. 1987; Ouellet 1980).

2.3 The Bioarchaeology of Osteoporosis

2.3.1 Bone Growth, Development, and Maintenance

To properly understand how osteoporosis and osteomalacia impact the integrity of the skeleton, it is essential to understand how normal bone forms, remodels, and is maintained over time. The ossification of the skeleton begins during fetal development, where a cartilage template of the postcranial skeleton is gradually replaced by osseous tissue, primarily composed of calcium phosphate (Isojima and Sims 2021; Poole et al. 2004). During ossification, osteoblasts enter the cartilage model through the vascular network of the forming bone and rapidly deposit osteoid that mineralizes into mechanically weak woven bone (Poole et al. 2004; Scheuer and Black 2004, 18-31). Osteoblasts and osteoclasts then remodel woven bone into strong lamellar bone (Boskey 2010; Poole et al. 2004; Scheuer and Black 2004, 18-31). Vitamin D, thyroid hormones, and other growth factors contribute to the longitudinal growth of long bones, ending with the fusion of the epiphyseal growth plates, regulated by estrogen (Fraher and Watson 2004; Gascon-Barré 2004; Hendy 2004; Morris et al. 2012; Poole et al. 2004; Riggs et al. 1998; Wiren 2005). During growth, it is necessary for osteoblast activity to outpace osteoclastic resorption in order to form skeletal structures (Frost 2003; Gocha et al. 2019; Parfitt 2003,

2013). By adulthood, when the skeleton has finished most of its growth, osteoblasts and osteoclasts balance their activity to maintain the skeleton by continuously remodelling bone (Frost 2003; Gocha et al. 2019; Kostenuik 2013; Parfitt 2003, 2013).

Bone amount is adjusted over time as a response to external factors that may temporarily change bone mass. Bone adapts to the strain produced by muscles in order to maintain the strength required to resist the forces of habitual muscle use while maintaining a lightweight and proportional structure (Almeida et al. 2017; Frost 2003; Harada and Rodan 2003; Prestwood and Raisz 2004; Wallace et al. 2017; Yuan et al. 2017). This process is continuously adjusting a bone's strength and microarchitecture to adapt to changes in an individual's activity over their life course (Frost 2003; Gocha et al. 2019).

The most significant source of fluctuating bone amount is through regulating calcium homeostasis in the body (Hendy 2004). When blood calcium levels are low, the parathyroid gland secretes parathyroid hormone (PTH), which acts on the kidney to increase calcium uptake during blood filtration and stimulates the conversion of 25-hydroxyvitamin D to its active form, 1 α ,25-dihydroxyvitamin D₃ (1,25D) (Anderson and Atkins 2008; Brickley et al. 2020; Lanske et al. 1999; Manolagas 2000; Minisola et al. 2021; Morris et al. 2012; Ringe 2020). This active form of vitamin D facilitates the uptake of phosphorous and calcium from the small intestine, and stimulates skeletal resorption to release calcium from bone (Glorieux and Pettifor 2014; Morris et al. 2012; Nelson and Villa 2003; Ringe 2020). When blood calcium levels are too high, calcitonin produced by the thyroid stimulates osteoblasts to convert calcium from the blood into bone (Cianferotti et al. 2015; Hendy 2004).

2.3.2 Peak Bone Mass

Peak bone mass is the maximum amount of bone an individual accumulates in their lifetime (Matkovic and Landoll 2004). Peak bone mass is typically reached around the third decade of life, when skeletal growth has finished completely (Laurent et al. 2014; Matkovic and Landoll 2004; Stini 2003), and is an important factor in the later development of osteoporosis (Hernandez et al. 2003; Laurent et al. 2014; Matkovic and Landoll 2004; Streeter and Stout 2003).

It is estimated that 50%-85% of bone mass is determined by genetics, while the remaining bone mass is determined by external factors (Matkovic and Landoll 2004; Ralston and Crombrugghe 2006; Ralston 2010; Streeter and Stout 2003). The ability to consistently obtain and process nutrients essential to skeletal growth in adequate quantities is essential to achieving a high peak bone mass. The importance of calcium and vitamin D in skeletal growth is stated above, but other compounds such as phosphorous, magnesium, potassium, and zinc are also crucial to the process of bone apposition, and to maintaining bone mass and bone quality over time (Prentice and Bates 1993).

2.3.3 Sexual Dimorphism

Androgens and estrogens are vital components of skeletal growth, maturation, and maintenance (Almeida et al. 2017; Laurent et al. 2014; Poole et al. 2004). In males and females, estrogen is essential to longitudinal growth and epiphyseal fusion in the developing skeleton (Almeida et al. 2017; Börjesson et al. 2010; Poole et al. 2004). During adolescence to the start of

puberty, the physiological operation of estrogen on the skeleton is similar in males and females (Almeida et al. 2017; Poole et al. 2004; Singh et al. 2010). Estrogen is responsible for longitudinal bone growth and epiphyseal fusion during skeletal maturation (Almeida et al. 2017; Börjesson et al. 2010; Giguère and Rousseau 2000; Seeman 2002; Singh et al. 2010). After skeletal maturity is reached, estrogen continues to play a crucial role in the regulation of bone homeostasis by suppressing excessive osteoclast activity (Almeida et al. 2017; Giguère and Rousseau 2000). Estrogen regulates osteoclast and osteoblast activity through direct action on bone cells, and indirectly through the suppression of cytokines and hormones that stimulate the formation of osteoblasts and osteoclasts (Almeida et al. 2017; Giguère and Rousseau 2000; Kousteni 2010).

Once sexual maturity is reached, however, the actions of sex hormones differ between males and females, creating sexually dimorphic features in the skeleton (Almeida et al. 2017; Laurent et al. 2014; Seeman 2002). Elevated androgen levels, such as those found in males after puberty, are responsible for continuous periosteal bone growth and increased skeletal robusticity due to greater muscle mass, resulting in higher average peak bone mass in males compared to females (Almeida et al. 2017; Laurent et al. 2014; Seeman 2002; Wiren 2005). Conversely, periosteal bone accrual slows in females during puberty, and subsequent bone apposition occurs on the endocortical surface (Almeida et al. 2017; Laurent et al. 2017; Singh et al. 2010). The differences in bone apposition regulated by sex hormones result in a greater cross-sectional bone area in males and a narrower endosteal area in females (Almeida et al. 2017; Isojima and Sims 2021; Laurent et al. 2014; Singh et al. 2010).

2.3.4 Osteoporosis

Once an individual reaches their peak bone mass, the rate of bone resorption overtakes the rate of bone deposition during regular remodelling activity, resulting in decreased bone mass and increased fragility (Almeida et al. 2017; Frost 2003; Grynpas 2003; Guerri et al. 2018; Hernandez et al. 2003). The diagnostic thresholds for bone loss established by the WHO indicates that bone can be considered osteoporotic when the BMD is 2.5 standard deviations or more below the average BMD for healthy young women (World Health Organization 1998). However, the onset and rate of bone loss is not uniform, leading to different types of osteoporosis:

2.3.4.1 Type I Osteoporosis

Type I osteoporosis is marked by a sudden drop in estrogen levels during menopause, which occurs on average between the ages of 45 and 55 in females (Black and Rosen 2016; Hernandez et al. 2003; Matkovic and Landoll 2004; Prestwood and Raisz 2004; Riggs et al. 1998; Schultz 2003). Estrogen inhibits osteoclast activity, making it a crucial component of regulating regular bone turnover (Almeida et al. 2017; Giguère and Rousseau 2000; Laurent et al. 2014). The consequence of this dramatic decrease in estrogen is an increase in osteoclast activity, resulting in a sharp decline in bone mass (Almeida et al. 2017; Prestwood and Raisz 2004; Riggs et al. 1998).

2.3.4.2 Type II Osteoporosis

Type II osteoporosis is primarily age-related (Almeida et al. 2017; Schultz 2003). In advanced age, intestinal and renal calcium absorption becomes less efficient, causing the parathyroid to hyperactively release PTH in an attempt to regulate calcium homeostasis, stimulating continuous skeletal resorption (Almeida et al. 2017; Lanske et al. 1999; Prestwood and Raisz 2004 Riggs et al. 1998). Type II osteoporosis typically occurs around the age of 70, and is typically seen in females more often than males (Almeida et al. 2017; Riggs et al. 1998). Osteoporotic bone loss begins with the trabecularization of bone on the endocortical surface (Frost 2003; Isojima and Sims 2021). Due to the sexually dimorphic properties of bone explored in Section 2.3.3, males have a greater ability to resist endocortical trabecularization through the continuous outward expansion of periosteal bone that compensates for endocortical loss (Almeida et al. 2017; Isojima and Sims 2021; Seeman 2002; Szulc et al. 2006). Because periosteal bone apposition in females is limited, periosteal expansion cannot compensate for endocortical bone loss, resulting in cortical thinning, trabecularization of the cortical bone, and a decrease in bone strength (Seeman 2002; Szulc et al. 2006). However, male androgen levels decrease with age, leading to a steady decline in bone mass over time that can result in type II osteoporosis (Almeida et al. 2017; Laurent et al. 2014; Riggs et al. 1998).

2.3.4.3. Other Forms of Osteoporosis

Although osteoporosis is primarily an age-related condition, there are several factors independent of the aging processes of the body that can contribute to the development of

osteoporosis (Schultz 2003). Endocrine disorders, gastrointestinal disorders, metabolic disorders, and genetic conditions can all impact the development of the skeleton (Rauch and Glorieux 2004). During periods of inflammation or infection, the cytokines secreted from immune cells can stimulate osteoclast activity, resulting in accelerated bone loss (Brickley et al. 2020, pp. 168; Mundy 2007; Riancho and Brennan-Olsen 2017).

Insufficient mechanical loading and disuse can cause bone loss, as the bone no longer requires the strength to resist mechanical stimulus (Bauman and Cardozo 2013; Brickley et al. 2020, pp. 169-170; Frost 2003). Disuse during recovery from fractures or illness, paralysis, amputation, and sedentary lifestyles have all been linked to the development of osteoporosis (Bauman and Cardozo 2013).

	T-Score	Z-Score
Normal Bone Density	T>-1	Z>-1
Osteopenia	-1>T>-2.5	-1>Z>-2.5
Osteoporosis	T<-2.5	Z<-2.5

Table 2.1. T- and Z-score values for the diagnosis ofosteoporosis (World Health Organization 1998).

2.3.5 Evaluating Osteoporosis in Archaeological Remains

Clinical diagnoses of osteoporosis and estimates of the prevalence of osteoporosis in a population typically rely on measurements of bone mineral density (BMD) obtained from dual energy x-ray absorptiometry (DXA) scans of the lumbar spine, proximal femur, and hip joint

(Blake and Fogelman 2007; Schultz and Wolf 2019; Siris et al. 2012; Wright et al. 2014). The values for BMD are then compared against a young, healthy control population of the same sex and ethnicity (T-score), and against a control population of the same age (Z-score) using the diagnostic values provided by the World Health Organization (Table 2.1) (Blake and Fogelman 2007; Schultz and Wolf 2019; Siris et al. 2012; World Health Organization 1998).

Bioarchaeologists face challenges when attempting to apply clinical diagnostic techniques to skeletal material. When examining osteoporosis in skeletal remains, it is often unclear if modern control samples can be used for comparison with archaeological samples (Mays 2018; van Spelde et al. 2021). Imprecise age estimation methods, particularly in older individuals, create challenges for comparison with a cohort group of the same age, and missing, incomplete, or damaged skeletal elements limit the usefulness of evaluating the prevalence of osteoporosis using the clinical methods described above (Agarwal 2012; Brickley and Buckberry 2015; Milner and Boldsen 2012). Instead, bioarchaeologists utilize a number of techniques to examine patterns of bone loss in a given sample. Bioarchaeologists continue to use measures of BMD, but also utilize measurements of element width, medullary width, cortical thickness, cortical area, cross-sectional shape, trabecular number, thickness, spacing, and length, and more, to determine patterns of bone loss in archaeological samples (Agarwal 2019; van Spelde et al. 2021). These techniques have utilized elements across the skeleton, including the femur, tibia, radius, lumbar vertebrae, ribs, and metacarpals (van Spelde et al. 2021).

Metacarpal radiogrammetry is a technique often used by bioarchaeologists to assess patterns of bone loss in archaeological samples using the second metacarpal (MCII). Metacarpal radiogrammetry originated as a technique for clinically diagnosing osteoporosis through the

quantification of cortical bone thickness (Barnett and Nordin 1960; Guerri et al. 2018; Meema and Meema 1987). The second metacarpal was chosen by clinicians due to its visibility in radiographs of the hand, as it is the largest bone in the hand, and mostly unobscured by soft tissue and the position of other bones in the hand and wrist (Barnett and Nordin 1960; Ives and Brickley 2004). When compared to techniques utilizing the quantification of bone amount in other elements, such as the femur and hip joint, the distal radius, and the lumbar spine, it was shown that the second metacarpal is a comparable element for analysis (Ives and Brickley 2005).

Traditional metacarpal radiogrammetry quantifies cortical bone thickness through a "cortical index," a standardized percentage that represents the width of cortical bone in relation to the total diaphyseal width at the mid-shaft of the bone (Barnett and Nordin 1960; Gilmour et al. 2021; Ives and Brickley 2004; van Spelde et al. 2021). Due to the simplicity and accessibility of this technique, it has been used by bioarchaeologists to explore patterns of bone amount in archaeological samples. Metacarpals tend to survive well in archaeological contexts (Waldron 1987), and radiograph equipment is easily accessible, fast, and inexpensive. These factors make metacarpal radiogrammetry a popular choice for assessing age and sex trends in bone amount, both within and between samples. A detailed procedure on the radiogrammetric methods utilized in this study are provided in Section 3.2.3.

2.4 Vitamin D Deficiency, Osteomalacia, and Osteoporosis

Vitamin D plays a crucial role in the development and maintenance of the skeleton. Vitamin D is also responsible for the formation of *good-quality* bone (Agarwal 2021; Arnold 1973; Brickley et al. 2020, pp. 129-163). Bone quality is determined by the microarchitecture of the bone, such as the organization of bone collagen, micro-damage, and proper mineralization (Arnold 1973; Brickley et al. 2020, pp. 129-163; Grynpas 2003; Karsdal et al. 2008; Seeman and Delmas 2006; Siris et al. 2012). Osteomalacia is a condition characterized by poor-quality bone, which causes soft bone that is susceptible to fracturing and bending (Bhan et al. 2018; Brickley et al. 2020, chapter 6; Minisola et al. 2021). The main cause of poor bone quality is vitamin D resistance or deficiency, interrupting the process of normal bone metabolism (Gascon-Barré 2004; Minisola et al. 2021).

Vitamin D is primarily acquired through the exposure of the skin to UVB rays from the sun. In the skin, the compound 7-dehydrocholesterol is formed, which is converted to vitamin D₃. Latitude and the solar zenith angle impact the amount of UVB that reaches the skin; locations near the equator receive the highest amount of UVB, while locations further north and south receive less UVB (Chaplin and Jablonski 2013; Holick 2007; Huotari and Hertzig 2008; Lips 2001; Mithal et al. 2009; Uday and Högler 2019). Individuals who live in areas that do not receive adequate UVB to produce vitamin D must supplement their vitamin D intake from dietary sources. Vitamin D obtained from food, such as fish and fortified dairy, is absorbed in the small intestine (Brickley et al. 2020, pp. 83-86; Gascon-Barré 2004; Morris 2012). Vitamin D obtained from both sunlight and diet is converted in the liver from vitamin D₃ to inactive 25-hydroxyvitamin D. The conversion to its active form, 1,25-D, takes place in the kidney as a

component of maintaining calcium homeostasis, as outlined in Section 2.3.1. It is important to note that although osteomalacia is most frequently caused by dietary and cutaneous vitamin D deficiency, it is possible to develop osteomalacia as a consequence of other pathological conditions (Brickley et al. 2020, pp. 87-89). Gastrointestinal disorders, liver and kidney damage, genetic mutations, hyperparathyroidism, fluorosis, tumors, and many more conditions can impact the ability to absorb and metabolize the components necessary for proper skeletal mineralization (Brickley et al. 2020, pp. 87-89; Minisola et al. 2021).

Diagnosing osteomalacia in clinical settings primarily requires measures of serum 25hydroxyvitamin D levels and histomorphometric analyses of bone biopsies (Bhan et al. 2018; Minisola et al. 2021). There is no consensus on the exact concentration of serum 25hydroxyvitamin D that indicates osteomalacia, but biomedical studies typically indicate that vitamin D deficiency occurs when serum levels fall in the range of 52 to 72 nmol per litre, and less than 50 nmol per litre indicates severe deficiency and risk of osteomalacia (Bischoff-Ferrari et al. 2006; Holick 2007; Malabanan et al. 1998). Bone biopsies rely on the histomorphometric markers of delayed mineralization and measures of osteoid thickness (Bhan et al. 2018; Need et al. 2007; Uday and Högler 2019). In ostomalacic bone, osteoid fails to properly mineralize during modelling and remodelling, resulting in a high ratio of osteoid to bone (Priemel et al. 2010). In histomorphometric evaluations, osteomalacia is clinically diagnosed when osteoid volume exceeds 5%, osteoid thickness exceeds 12.5µm, and a mineralization lag time of over 100 days is observed (Bhan et al. 2018; Minisola et al. 2020; Need et al. 2007; Uday and Högler 2019).

When normal remodeling is disrupted due to osteomalacia, layers of bone that should ossify fail to mineralize or do not mineralize completely, resulting in defects in the cortical and

trabecular bone that decrease the quality and structural integrity of the bone (Bhan et al. 2018; Glorieux and Pettifor 2014; Priemel et al. 2010). Meanwhile, insufficient vitamin D impairs calcium absorption, stimulating osteoclast activity to release calcium into the blood, resulting in healthy and properly mineralized bone becoming resorbed (Glorieux and Pettifor 2014; Minisola et al. 2021; Morris et al. 2012). Defectively mineralized bone and the resorption of healthy bone results in weakened bone that becomes more susceptible to fracturing (Bischoff-Ferrari et al. 2006; Minisola et al. 2021). Therefore, while osteoporosis and osteomalacia are distinct pathological conditions, the factors listed here suggest they are linked through many of the same physiological processes.

Chapter 3: Materials and Methods

3.1 Materials

The sample in this study is composed of individuals from two historical French-Canadian cemetery collections, Pointe-aux-Trembles and Sainte-Marie, currently curated in the Department of Anthropology at the University of Montréal. Excavations were undertaken for the purpose of archaeological intervention and data rescue due to urban development at both sites (Ethnoscop 2006, 2016).

3.1.1 Pointe-aux-Trembles

The village of Pointe-aux-Trembles is located on the northeast shoreline of the Island of Montréal along the St. Lawrence River (Figure 3.1) and was officially founded in 1674 (Ethnoscop 2016). The village was initially constructed as a French Fort to protect the colonists of the Island of Montréal and to control the St. Lawrence River. Following a succession of Iroquois attacks on the colonial farms, a fortified palisade was constructed to protect families. The construction of a mill and church within the walls transformed the fort into a village. Soon after, the *Chemin du Roy* ("King's Road") was constructed to connect Montréal, Québec City, and the settlements on the north bank of the St. Lawrence River. This road passed through Pointeaux-Trembles, which increased traffic to the village (Desjardins 2010; Robichaud 2012).



Figure 3.1. Map of Montréal, courtesy of Google (2023a). Pointe-aux-Trembles is indicated by the star.

The cemetery of the local church, Saint-Enfant-Jésus parish, was in use from 1709-1843 (Table 3.1), and by the end of this period, the cemetery contained 2500 burials. This cemetery was excavated in 2014 by Ethnoscop Ltd. during the process of developing and modernizing structures in the area. A trench of approximately 12m by 2m was excavated and included 63 graves, some of which contained multiple individuals. Graves were oriented in a variety of directions and organized in clusters, potentially representing family plots (Ethnoscop 2016). The excavation revealed graves stacked atop one another in three layers, representing three phases of use of the cemetery until its closure in 1843. The oldest layer (coded 7A11) corresponds with the earliest uses of the cemetery in 1709. The middle layer (coded 7A9) represents the first addition of new earth to the cemetery to create more space for burials. Cultural artifacts
M.A. Thesis- R. Christenson, McMaster University - Anthropology

indicate this layer of the cemetery was in use approximately from the mid-1700s to the early 1800s. The final layer (designated 7A2) was in use until 1843. Twenty-three adults were recovered from this excavation, and 18 of these adults were included in this study. The criteria for inclusion in this study are outlined in Section 3.1.3.

Table 3.1. Dates of cemetery	y use at Pointe-aux-Trembles and Sainte-Marie.

	Pointe-aux-Trembles	Sainte-Marie
Years of Use	1709-1843	1748-1878

3.1.2 Sainte-Marie

The municipality of Sainte-Marie is located approximately 50 kilometres south of Québec City, on the bank of the Chaudière River (Figure 3.2). The first settlers arrived in Sainte-Marie in 1738, and the first interments occurred in 1748 (Table 3.1), six years before the first chapel was erected on the property (Ethnoscop 2006). The cemetery was in use until the 1870s, having undergone numerous expansions in both area and height. By 1832 the cemetery reached its full capacity, and earth was added to accommodate new graves in 1840. In 1864 the cemetery became full again and was extended in length, but less than a decade later in 1873 the cemetery was raised with another layer of earth. Finally, in 1878, the cemetery was permanently closed, containing nearly 7000 burials and occupying nearly 18,000 square feet of land.

The graves in the cemetery were not relocated after its closure, leaving thousands of graves in situ beneath what is presently the parking lot of the modern Sainte-Marie Catholic Church. Excavation of the Sainte-Marie cemetery was undertaken in 2003 and 2004 by

Ethnoscop Ltd. in response to redevelopment efforts on the parish property. Nine plots with dimensions of approximately 2m by 2m were excavated. A total of 62 individuals were removed from these plots, including 29 adults. Twenty-five of these adults were included in this study. The criteria for inclusion in this study is outlined in Section 3.1.3.



Figure 3.2. Map of Québec, courtesy of Google (2023b). Sainte-Marie is indicated by the star.

3.1.3 Criteria for Inclusion

The sample of interest for this study includes adult males and females with at least one second metacarpal (MCII) with 1.9cm of the diaphyseal surface intact. Priority was given to the right MCII, but the left was used if the right MCII was missing, fragmentary with less than 1.9cm of the diaphysis present, or altered by taphonomic change beyond the parameters established by Gilmour et al. (2021). Adults with one MCII were then assessed for age at death and sex.

Individuals were further excluded in cases where sex was indeterminate and age estimates were ambiguous. The demographic information for the final study sample is summarized in Tables 3.2 and 3.3, and Appendix A.

Table 3.2. Sample distribution by sex.

	Number of Females	Number of Males	Total
Pointe-aux-Trembles	12	6	18
Sainte-Marie	15	10	25
Total	27	16	43

Table 3.3. Sample distribution by age.

	Age 15-29	Age 30-49	Age 50+	Total
Pointe-aux-Trembles	9	5	4	18
Sainte-Marie	10	5	10	25
Total	19	10	14	43

3.2 Methods

3.2.1 Age Estimation

Age at death estimation was undertaken using a multifactorial approach, in order to counter the limitations of each method and to gather as much age information as possible from highly fragmented remains. Features of the auricular surface (Meindl and Lovejoy 1989) and pubic symphyseal surface (Brooks and Suchey 1990) were prioritized. In addition to these features, unfused long bone epiphyses (Buikstra and Ubelaker 1994), vertebral fusion (Albert et al. 2010; Albert and Maier 2013), and the appearance of the sternal rib ends (İşcan et al. 1984) were scored, particularly in cases where the features of the pelvis were not visible. Age and sex

M.A. Thesis- R. Christenson, McMaster University - Anthropology

estimations were recorded using the form in Appendix B1. After estimating age at death, individuals were placed within three age categories: 15-29, 30-49, 50+.

Individuals assigned an age of 50 years or older using the previous techniques were assessed again for indicators of advanced age through transition analysis, which has been tested as an effective technique for assessing age in individuals above 50 years of age (Fojas et al. 2018; Boldsen et al. 2002; Milner and Boldsen 2012). Transition analysis relies on the development of entheseal changes in the sacroiliac joint and its proximate muscle attachments sites and the dorsal margin of the symphyseal surface as the main sites of age-related change (Bolsden et al. 2002). The development of iliac exostoses on the posterior aspect of the auricular surface and the breakdown of the dorsal margin of the pubic symphysis are the main feature of age-related entheseal change in older individuals, indicating the individual is likely over 80 years of age (Bolsden et al. 2002).

3.2.2 Sex Estimation

Sex estimations relied mainly on the morphological features of the pelvis, such as the greater sciatic notch, the preauricular sulcus, and features of the subpubic region (Buikstra and Ubelaker 1994, pp. 16-19). Sexually dimorphic markers of the skull (Acsadi and Nemeskeri 1970) were only be utilized if the pelvic features are not visible (Appendix B1).

3.2.3 Metacarpal Radiogrammetry

3.2.3.1 Metacarpal Radiogrammetry and the ROI Method

Traditional metacarpal radiogrammetry measures cortical bone width relative to the total width of the metacarpal midpoint to calculate a cortical index (Barnett and Nordin 1960; Gilmour et al. 2021; Ives and Brickley 2004; van Spelde et al. 2021), but recent studies have expanded this traditional methodology to quantify cortical bone within a larger area (Gilmour et al. 2021). This study utilizes a newly developed method of quantifying cortical bone amount developed by Gilmour et al. (2021). Instead of measuring the midpoint width of the metacarpals to calculate a cortical index, this new technique calculates the area of cortical bone within a standardized region of interest (ROI) on the diaphyseal surface. This value is referred to as the cortical area index (CAl_{ROI}).

Second metacarpals were radiographed in groups of five using Vidisco portable digital radiography equipment at the University of Montréal. Metacarpals were secured in anteroposterior orientation on a sheet of plastic air bubble packaging and spaced one inch apart. The scale of each image was calibrated using a 23.88mm Canadian 25-cent coin.

To collect the measurements for the CAI_{ROI} calculations, the procedure outlined by Gilmour et al. (2021) was followed using ImageJ software (Version 1.53). A 1.9cm x 1.9cm square was drawn using the ROI selection tool and placed over the narrowest portion of the diaphyseal surface. If necessary, the contrast of the image was adjusted to increase the visibility of the cortical and trabecular structures within the ROI. Using the oval selection brush tool, the total bone area, including the cortical area and the medullary area, were selected and used as the total area (TA) value. The cortical bone area (CB) was then isolated and saved as the CB value (Appendix B2). Using these values, the cortical area index in the region of interest was calculated with the following formula:

$$CAI_{ROI} = 100 \times \frac{CB}{TA}$$

Initial measurements were taken in the field using a laptop with a 14-inch screen and a resolution of 1920x1080P, in order to obtain preliminary CAI_{ROI} values, allowing rib samples to be chosen for micro-CT assessment across a range of bone amounts. Measurements were taken a second time using a 27-inch monitor with a resolution of 3840x2160P to obtain more precise values.

3.2.3.2 Statistical Analysis

The data from both Pointe-aux-Trembles and Sainte-Marie was pooled, based on the assumption that the demographic information and conditions of each site were comparable. Utilizing historical sources, it was noted that there were no major differences in social or economic status that would result in vastly different nutrition patterns and population health (Ethnoscop 2006, 2016; Le Moine 1876; Lucas 1901). Due to this, it was deemed appropriate to pool the raw data. Pooling these numbers is beneficial for the subsequent statistical analyses, as they create a larger sample size.

Statistical tests were carried out on SPSS (Version 28.0.1.1) and considered significant at a *p*-value of \leq 0.05. Before performing any comparative tests, the Shapiro-Wilk test for normality

of distribution was carried out for each age and sex category. The Shapiro-Wilk test was chosen due to the small sample sizes (n<20) in each category (Shapiro and Wilk 1965). All categories were determined to be normally distributed, therefore all subsequent comparative tests must be parametric tests.

The first test assessed the statistical differences between CAI_{ROI} values between males and females in the overall sample. Given that cortical bone area is sensitive to sex hormone levels, as outlined in Section 2.3.3, a significant difference between sexes would require the relationship with age to be considered separately for males and females, whereas no statistical difference would allow males and females to be considered together in the test for age and CAI_{ROI}.

Student's t-test was used within each age division to assess the significance of the CAI_{ROI} difference between males and females. The t-test is a parametric test that evaluates the statistically significant differences between the means of two normally distributed samples (Dakhale et al. 2012; Suchmacher and Geller 2021 pp 113-123). One-way ANOVA was then used to determine statistical differences in CAI_{ROI} values between each of the total age categories (15-29, 30-49, 50+). Results for both the t-test and ANOVA tests were considered statistically significant at p<0.05. Finally, age and CAI_{ROI} were compared in the total sample using Pearson's correlation coefficient (*r*) and linear regression analysis (r^2).

3.2.3.3 Comparative Analysis

An extensive literature review was undertaken to identify other bioarchaeological studies that provided data on age and sex trends in bone amount in archaeological samples for comparative data. A systematic search of scientific databases such as PubMed, JSTOR, and ScienceDirect using the keywords "bone mineral density," "cortical bone loss," "age-related bone loss," "osteoporosis," and "paleopathology" was undertaken. Titles and abstracts of articles were scanned for relevance. Articles were required to report skeletal data for both males and females, and had to report bone amount for each age category established in the methodology, rather than reporting overall trends. The results are reported in Appendix C.

3.2.4 Micro-CT Analysis

3.2.4.1 Sample Selection

Rib samples were collected from ten individuals for a case-study analysis of osteomalacia in Pointe-aux-Trembles and Sainte-Marie. To examine the relationship between bone amount and vitamin D deficiency, individuals with high, low, and average CAI_{ROI} values were systematically selected to represent the total CAI_{ROI} range. In addition, individuals with macroscopic skeletal lesions highly consistent with osteomalacia, such as pseudofractures in the pelvis or alterations in rib morphology (Brickley et al. 2020 pp. 117-121), were also chosen for further analysis.

3.2.4.2 Micro-CT Imaging

Rib samples were chosen for micro-CT analysis for several reasons. A study of bone survival rates determined that 40%-59% of ribs survive archaeological contexts (Waldron 1987), meaning it is likely that an individual will have at least some rib fragments that can be chosen for analysis. Ribs have a turnover of approximately 5 years, compared to other bones in the skeleton that have turnover rates or 10-20 years (Jørkov et al. 2009). The fast turnover rate in ribs means that observed defects are more representative of skeletal health at the time of their death than defects observed in other skeletal elements.

The chosen samples were analyzed using the Nikon XTH-225ST micro-CT scanner at the Ancient Images Lab of the Museum of Ontario Archaeology in London, Ontario. The rib samples were mounted in a Styrofoam cup and stabilized with green florist foam. Samples were scanned using a low kVp technique (80kVp, 173A), a molybdenum target to maximize contrast, and a voxel size of 10.5 µm. Each sample was scanned for 53 minutes, with a total of 3141 2D x-ray images taken as the mounted sample rotates in the scanner, at a rate of 1 projection per second. The 3D reconstruction was completed using X-Tec CT-Pro, V. 4.4. The files generated from the reconstruction were uploaded to Dragonfly (Version 2021.3 Build 1087) where each 2D slice and the 3D image could be visually analyzed.

3.2.4.3 Slice Analysis and Scoring

A range of histological features suggestive of osteomalacia have been explored by researchers (Section 2.4), although some are invisible in micro-CT scans. The most distinctive

histological indicator of osteomalacia is defective mineralization along cement lines (DCL). These defects form when new layers of bone fail to ossify completely during life, leaving distinct crescent-shaped structures of bone that appear partially or completely disconnected from other bone structures (Brickley et al. 2007; Priemel et al. 2010; Welsh et al. 2020). Areas of incomplete mineralization, which appear darker than mineralized bone in micro-CT scans, can also be indicative of defective mineralization. However, because bone is constantly modelling and remodelling (Frost 2003), patches of incomplete mineralization may be visible in those without osteomalacia. Areas of incomplete mineralization are noted, but not given the same scoring weight as DCLs.

Score	Description
0	No clear defects.
1	Visible defects are small in size. Defective cement lines are visible but have not completely separated new and old bone.
2	Visible defects are more pronounced. Defective cement lines span a large area and may show partial separation between new and old bone.
3	Defects are clearly visible and seen consistently in the scan. Defective cement lines may show the complete or marked partial separation of new and old bone surfaces and may span a large area.

To assess the prevalence and consistency of DCLs throughout the rib, five equally spaced slices, spaced 250 scans apart, were chosen from within a 1cm region of interest on the shaft of the rib. From these five images, it was possible to note not only the appearance and number of mineralization defects in a single slice, but the consistency with which they occurred throughout the sample. Utilizing several slices in this region, the chances of misidentifying diagenesis or taphonomic change as a pathological defect is reduced.

Each slice was visually analyzed and assigned a score from 0 to 3 based on the presence, appearance, and size of the DCLs (Table 3.4). Slices with DCLs that were improperly mineralized were assigned an extra half-point to the slice score. Visual depictions of the appearance of each defect type are represented in Figure 3.3-3.4. The scores from each slice were then added together to assign an overall defect status to the individual (Table 3.5).



Figure 3.3. Appearance of defective cement lines with a score of 1 (Image A), 2 (Image B), and 3 (Image C). Scale bars indicate 1mm.



Figure 3.4. Appearance of incomplete mineralization associated with defective cement lines. The cement lines are scored as a 1 (Image A), 2 (Image B), and 3 (Image C). Scale bars indicate 1mm.

Total Defect Score	Overall Defect Category
0.0-3.5	No significant defects
4.0-7.5	Mild defects
8.0-11.5	Moderate defects
12.0+	Severe defects

Table 3.5. Defect category derived from total defect score.

3.2.4.4 Interpretations Through Osteobiography

Due to the small sample size, it was not possible to perform a statistical analysis of defect scores in relation to age, sex, or CAI_{ROI}. Instead, the defect scores were interpreted through an osteobiographical approach, exploring the experiences of the individual situated within the general characteristics of the sample, and the broader social and cultural context in which the sample lived.

Chapter 4: Results

4.1 Introduction

This chapter outlines the results of this study, beginning with an analysis of the relationship between age, sex, and the total cortical area of the mid-diaphyseal region of interest (CAI_{ROI}) values obtained using the radiogrammetric procedure outlined in Section 3.2.3 (Appendix D). Following the CAI_{ROI} data analysis, observational results of the ten rib micro-CT scans are reported. Finally, the CAI_{ROI} data and the micro-CT results will be summarized and compared in the ten individuals selected for rib scans.

4.1.1 Sample Size and Distribution

The total sample includes 43 individuals; 18 individuals were chosen from Pointe-aux-Trembles, and 25 individuals were chosen from Sainte-Marie. Of these 43 individuals, rib samples were chosen from 10 individuals for micro-CT scanning for an observational analysis of mineralization defects associated with osteomalacia likely caused by vitamin D deficiency.

4.2 Metacarpal Radiogrammetry

4.2.1 Sex and CAI_{ROI}

As explored in Section 2.3.3, the processes of bone accumulation and age-related bone loss follow sex-related patterns, particularly as estrogen levels fluctuate throughout a female's

lifespan. All data was determined to be normally distributed (Appendix E1). In order to determine if further calculations of CAI_{ROI} should be separated by sex, student's t-test was used to test for statistical differences in CAI_{ROI} values by sex. A test in the 15-29 age category (*f*=11, *m*=8), the 30-49 age category (*f*=8, *m*=2), the 50+ age category (*f*=8, *m*=6), and a test for the total sample (*f*=27, *m*=16) were carried out (Appendix E2).

In each of the four analyses, there were no statistically significant differences in the CAI_{ROI} values between males and females (p>0.05). Because there are no differences between the sexes in any age group, it is appropriate to combine male and female data for further analyses.

4.2.2 Age and CAI_{ROI}

The 15-29 age category had the widest range of CAI_{ROI} values of all three age groups (Appendix E3), and contained the individuals with both the highest overall CAI_{ROI} value (2E13) and the lowest overall CAI_{ROI} value (7A9-S45). The CAI_{ROI} ranges become narrower and the median values decrease with age (Figure 4.1). To statistically analyze the relationship between CAI_{ROI} and age observed in Figure 1, one-way ANOVA and Pearson's correlation coefficient (*r*) were applied (Appendix E4).



Figure 4.1. Box plot of cortical area index (CAI_{ROI}) in relation to age (*n*=43).



Figure 4.2. Scatter plot depicting the relationship between age and cortical area index (CAI_{ROI}) (*n*=43).

M.A. Thesis- R. Christenson, McMaster University - Anthropology

To assess the relationship between age and CAI_{ROI} , one-way ANOVA was used to compare the CAI_{ROI} values of each age group for statistically significant differences. The results from each test reveal that there are no statistically significant differences in the average CAI_{ROI} values between any of the age categories (p>0.05).

Pearson's correlation coefficient was applied to the total sample to assess the strength of the correlation between age and CAI_{ROI} . Pearson's correlation coefficient represents a number between -1 and 1, where the closer the *r* value is to 0, the weaker the relationship between the variables. An *r* value of -0.085 was calculated for the relationship between age and CAI_{ROI} , indicating an extremely weak negative correlation between age and CAI_{ROI} for this sample (Figure 4.2), although this was determined not to be statistically significant.

4.3 Micro-CT Analysis

Rib samples from ten individuals were chosen from the total sample for micro-CT scanning. The chosen sub-sample consisted of seven individuals from Pointe-aux-Trembles (f=5, m=2) and three individuals were chosen from Sainte-Marie (f=2, m=1). Mineralization defects were scored on the scale described in Section 3.2.4 and reported in Appendix F.

4.3.1 Pointe-aux-Trembles

4.3.1.1 7A2-S2

The age and sex assessment of individual 7A2-S2 estimated that this individual is a female in the age category of 30-49, and the radiogrammetric assessment of their metacarpal indicated a CAI_{ROI} of 69.045. The micro-CT scan of individual 7A2-S2's left rib indicates

significant fungal tunnelling and taphonomic damage, rendering it difficult to view potential mineralization defects. The trabecular spacing is inconsistent throughout the rib, transitioning from a healthy microarchitectural network to areas of sparsely arranged trabeculae. The cortical bone is thin, particularly in the areas where there are many trabecular structures. There are several instances of defective cement lines in the rib, particularly in the areas with a strong trabecular network (Figure 4.3). A total score of 6.5 was assigned to this individual, placing them in the mild defect category.

4.3.1.2 7A2-S11

Individual 7A2-S11 is estimated to be a female in the 30-49 age category, with a CAI_{ROI} value of 32.956. The left rib fragment chosen for scanning is damaged on its external surface, and the images of the internal structures show evidence of fungal tunnelling and several soil deposits within the trabecular structure. The trabeculae are sparse and the cortical bone is thin, particularly along the lateral portion of the rib. There are several small, consistent cases of defective cement lines and improper mineralization throughout the scan. This individual was assigned a score of 10.5, placing them in the moderate defect category.

4.3.1.3 7A2-S20

Individual 7A2-S20 is estimated as a female within the 15-29 age category. Their CAI_{ROI} was determined to be 29.382. The trabeculae of the rib are extremely sparse and the cortical area is thin. There are numerous areas of improper mineralization within the cortical bone, and

a small number of defective cement lines (Figure 4.4). The sum of defect scores is 5.5, assigning them to the mild defect category.



Figure 4.3. Defective cement lines (DCL) in individual 7A2-S2, slice 1300. White arrows indicate stage 1 DCLs, orange arrows indicate stage 2 DCLs.

Figure 4.4. Defective cement lines (DCL) in individual 7A2-S20, slice 1350. White arrows indicate stage 1 DCLs.

4.3.1.4 7A9-S38

The age and sex estimates for individual 7A9-S38 determined they are likely male within the 15-29 age category, and their CAI_{ROI} value is 67.181. The rib scan for this individual shows considerable taphonomic damage and soil infilling; however the cortical area and trabecular network are healthy. Any potential mineralization defects are ambiguous, and do not appear consistently throughout the scans, receiving a defect score of 0.

4.3.1.5 7A9-S46

Individual 7A9-S46 is likely a female, in the 50+ age category, with a CAI_{ROI} of 49.569. The scan of the rib shows extensive fungal damage to both the thinned cortical bone, which shows extreme evidence of endocortical trabecularization, and sparse trabecular structures. In portions of the bone that are not damaged, several areas with long defective cement lines are visible throughout the rib (Figure 4.5). This score assigned to this individual was 6.5, assigning them to the mild defect category.

4.3.1.6 7A11-S19

This individual was estimated to be a male in the 50+ age category. The radiogrammetric assessment of their metacarpal indicated a CAI_{ROI} of 43.313. The trabecular structure is sparse and the cortical bone is damaged by extensive fungal tunnelling and other taphonomic damage. Despite these factors limiting the visibility of potential defects, several mineralization defects are consistently visible throughout the scan. This individual was assigned a total score of 12.5, indicating severe defects.



Figure 4.5. Defective cement lines (DCL) in individual 7A9-S46, slice 450. White arrows indicate stage 1 DCLs. The white areas within the red circle are soil inclusions within the rib.

4.3.1.7 7A11-S42

Individual 7A11-S42 is estimated to be a female within the 15-29 age category, with a

CAI_{ROI} value of 27.813. The internal structures of the scanned rib fragment are extensively

damaged, and there are several large soil deposits within the bone. The intact trabeculae are thin and sparse, and the cortical bone is extremely thin. Small areas of DCLs throughout the scans resulted in a score of 4.5, placing them in the mild defect category.

4.3.2 Sainte-Marie

4.3.2.1 2B9

This individual is likely a female in the 50+ age category. Their CAI_{ROI} value is 31.012, and the scan of their rib had a defect score of 1.0, indicating no significant mineralization defects. The trabeculae, while sparse, are relatively thick, and the cortical bone appears healthy.

4.3.2.2 2E13

Individual 2E13 was assessed as a young female within the 15-29 age group. The radiogrammetric assessment of their metacarpal indicated a CAI_{ROI} of 73.856. Despite some taphonomic damage to the internal structures of the rib, the trabecular network and cortical bone appear healthy. The scan shows several large cases of defective cement lines that persist throughout the rib, and several cases of improper mineralization (Figure 4.6a, b). This individual was assigned a defect score of 14.5, indicating severe defects.

4.3.2.3 2G4

Individual 2G4 is a probable male within the 15-29 age category, with a CAI_{ROI} of 43.449. Despite slight fungal damage and cracking in the cortical bone, the internal structures of the rib are easily visible. The trabeculae are sparse in some areas throughout the rib, but the cortical bone is healthy and has undergone minimal thinning. This individual received a total defect score of 2.0, indicating there are no significant defects present.



Figure 4.6. Defective cement lines (DCL) in individual 2E13, slice 1300 (A) and 1800 (B). White arrows indicate stage 1 DCLs, orange arrows indicate stage 2 DCLs, and blue arrows indicate stage 3 DCLs.

4.3.3 Mineralization Defects and CAI_{ROI}: Comparison and Summary

The CAI_{ROI} data and the mineralization defect scores for the ten individuals chosen for micro-CT scanning are compared in Table 4.1. Several individuals with high CAI_{ROI} values showed substantial evidence of mineralization defects, particularly individual 2E13. Individual 2E13 is a

female in the 15-29 age category, and has the highest overall CAI_{ROI} in the total sample;

however, the scanned rib from 2E13 shows the most severe and numerous mineralization

defects of the entire sub-sample. In contrast, individual 2B9 is an older female in the 50+ age

category with one of the lowest overall CAI_{ROI} values, and shows no clear evidence of

mineralization defects in the micro-CT scan. There does not appear to be a clear relationship

between CAI_{ROI}, age, and the presence or severity of mineralization defects in this sample.

Table 4.1. Summary of age, sex, cortical bone area (CAI_{ROI}), and mineralization defect scores (n=10).

	Age	Sex	CAI _{ROI}	Mineralization
				Defects
7A2-S2	30-49	F	69.045	6.5
7A2-S11	30-49	F	32.956	10.5
7A2-S20	15-29	F	29.382	5.5
7A9-S38	15-29	М	67.181	0.0
7A9-S46	50+	F	49.569	6.5
7A11-S19	50+	М	43.313	12.5
7A11-S42	15-29	F	27.813	4.5
2B9	50+	F	31.012	1.0
2E13	15-29	F	73.856	14.5
2G4	15-29	М	43.449	2.0

Chapter 5: Discussion

5.1 Introduction

The previous chapter presented a detailed statistical analysis of the calculated cortical area index values (CAI_{ROI}), determined from the second metacarpals of the individuals Pointe-aux-Trembles and Sainte-Marie. Following the statistical analysis, micro-CT images of ribs from the 10 selected individuals were analyzed individually and scored for evidence of mineralization defects associated with vitamin D deficiency. In this chapter, patterns of age-related bone loss at the sample level in Pointe-aux-Trembles and Sainte-Marie will be discussed and compared to expected trends observed in clinical data and to other bioarchaeological studies of the same nature. The results of the rib micro-CT assessment will be discussed in-depth through individual osteobiographical accounts. This chapter will close with a discussion of osteoporosis and osteomalacia as social history in historical French-Canada.

5.2 Metacarpal Radiogrammetry

Using the metacarpal radiogrammetry method developed by Gilmour et al. (2021), the current study revealed no relationship between age and cortical bone amount in the study sample, nor were there significant differences in cortical bone amount between sexes. These results do not follow the expected trends outlined in clinical sources using living individuals. Clinical data shows that bone reaches its peak density near the third decade of life and begins to decline shortly after this period (Agarwal 2021; Compston et al. 2019; Keaveny and Yeh 2002; Manolagas 2000; Prestwood and Raisz 2004). Medical literature also documents sex-based

M.A. Thesis- R. Christenson, McMaster University - Anthropology

differences in bone accumulation and total mass, with males reaching higher peak bone levels than females, while females lose more extreme bone levels in older age due to menopause (Black and Rosen 2016; Isojima and Sims 2021; Laurent et al. 2014; Manolagas 2000; Matkovic and Landoll 2004; Riggs et al. 1998). An extremely weak negative correlation between age and CAI_{ROI} was identified using Pearson's correlation coefficient (*r*=-0.085), indicating a slight relationship between bone loss and old age. However, this value was not statistically significant (p>0.05). It is unexpected that this study would find no age or sex-based discrepancies in CAI_{ROI} in the individuals from Pointe-aux-Trembles and Sainte-Marie.

5.2.1 Comparative Analysis

Many cautions must be taken when comparing data from archaeological samples to modern medical data. Small sample sizes (Wissler et al. 2022), and poor element preservation (Brickley and Buckberry 2015; de Boer and van der Merwe 2016; Gilmour et al. 2021) are the main limiting factors in bioarchaeological research. Therefore, other studies on bone loss in archaeological samples serve as effective sources of comparative data. Approximately 70 bioarchaeological studies investigating age and sex trends in bone were identified. Of the 70 studies, 13 papers considering a total of 1966 individuals met the inclusion criteria outlined in Section 3.2.3.3. Three trends were revealed from the results of the comparative studies: 1) Adherence to clinical age and sex trends, 2) Age-related bone loss with sex-based deviations from clinical data, 3) No relationship between age, sex, and bone loss. The results of this literature review are presented in Table 5.1.

Reference	Location and Sample Groups	Age-Related Bone Loss	Sex-Based Differences in Bone Amount	No Age or Sex Trends
Agarwal 2012	Rural England	\checkmark		
	Urban England	\checkmark	\checkmark	
Agarwal and Grynpas 2009	England	\checkmark	\checkmark	
Beauchesne and Agarwal 2014	Italy	\checkmark		
Beauchesne and	Italy Sample 1	✓	✓	
Agarwal 2017	Italy Sample 2	✓		
	Italy Sample 3	✓	✓	
Curate et al. 2013	Portugal	✓	✓	
Dewey et al. 1969	Sudan Sample 1			✓
	Sudan Sample 2	√	✓	
	Sudan Sample 3			✓
Ekenman et al. 1995	Sweden			~
Ericksen 1976	Southwest United States	~	✓	
	Midwest United States	✓	✓	
	Alaskan Coast	√	✓	
Glencross and Agarwal 2011	Turkey	\checkmark	✓	
Holck 2007	Norway Sample 1	✓		
	Norway Sample 2	\checkmark		
	Norway Sample 3	✓		
Poulsen et al. 2001	Denmark			✓
Van Gerven et al. 1969	Midwest United States	~	✓	
Zaki et al. 2009	Egypt	✓	✓	
Total		18	12	4

Table 5.1. Summary of observed age and sex trends in comparative studies.

In reviewing bioarchaeological literature from a wide range of spatial and temporal contexts, many studies commonly reported trends which followed the age and sex expectations established by clinical data. In a study by Glencross and Agarwal (2011), metacarpal radiogrammetry was used on the second metacarpals of 49 adults (f=27, m=22) from the

Neolithic community of Çatalhöyük, Turkey. Individuals were placed in three age groups similar to the categories established in this study, and it was ultimately found that the cortical bone amounts measured in this sample adhere to the age and sex trends set by clinical data.

Curate et al. (2013) examined 196 individuals of known age, sex, and ancestry from a rural community cemetery in central Portugal, occupied from the mid-19th to early-20th century. Dual X-ray absorptiometry of the proximal femur was used to measure the bone mineral density (BMD) of these individuals. The authors observed a significant decrease in BMD with age and a pronounced discrepancy in BMD between males and females, particularly in the older age groups.

Van Gerven et al. (1969) measured femoral cortical bone thickness in 43 individuals (f=23, m=20) excavated from a prehistoric Mississippian site, using roentgenographic x-ray images and direct measurements from femoral slices. Statistical analyses of these measurements show a significant decrease in cortical bone amount with age in the total sample, and substantial sex-based differences in bone amount, as females lost significantly more bone with age compared to males. A later study of three Indigenous archaeological samples by Ericksen (1976) utilized the same roentgenographic methods on the humerus and femora of these individuals. This study too found the same results as the above studies, consistent with clinical trends.

Bioarchaeological studies frequently report deviations in the sex-based trends established in clinical literature, particularly in female skeletons. While male samples typically adhere to the gradual progression of bone loss over time, the timing and ultimate trajectory of

M.A. Thesis- R. Christenson, McMaster University - Anthropology

changes in bone amount is considerably different in females in many studies compared to modern medical trends and their male counterparts. Agarwal and Grynpas (2009) measured vertebral bone loss in a British medieval assemblage and observed that while males follow the expected pattern of bone loss over time, the female cohort reached the lowest levels of bone by middle age (30-49), and did not undergo any significant loss following this initial drop. Zaki et al. (2009) also report an initial early reduction in BMD between the second and third decades of life in an assemblage of Egyptian skeletons, particularly in the female study sample.

Bioarchaeological studies of age-related bone loss frequently report no statistically significant differences in bone amount between sexes. In an Imperial Roman assemblage, Beauchesne and Agarwal (2014) found no differences in cortical bone index between males and females in any age category (18-29, 30-49, 50+). Similar results were reported by Holck (2007) in a study of BMD in prehistoric, Viking Age, and medieval samples in Norway. There were no significant differences in BMD between males and females in each of the three samples. This trend is also observed in Agarwal (2012) and Beauchesne and Agarwal (2017), among other papers.

The least-commonly observed results report no relationship between age, sex, and bone amount. Ekenman et al. (1995) radiographically determined BMD in 343 individuals from a medieval cemetery in Stockholm and found no changes in bone amount over time for males or females. Poulsen et al. (2001) estimated the BMD of 49 individuals excavated from a Danish Christian cemetery and found no change in BMD over time in males, and an *increase* in BMD with age in females. During the literature review, only one study reported results similar to those of this thesis. In a study of three agricultural communities from different time periods

(350 B.C.- A.D. 350, A.D. 350- A.D. 550, A.D. 550- A.D. 1300) in modern Sudan by Dewey et al. (1969), two of the study groups showed extreme fluctuations over time in both the male and female samples and ultimately showed no correlation between age, sex, and cortical bone amount.

5.2.2. Bone Amount as Bioculture

In Section 2.3, the biological processes responsible for the accumulation and loss of bone are outlined from a clinical perspective, with emphasis placed on the role of genetics and sex hormones as determinants of peak bone mass and age-related bone loss. Particular emphasis is placed on managing postmenopausal osteoporosis, as clinicians provide statistics and estimates that place females at a much higher risk of osteoporosis compared to males. For example, one in three women are likely to experience osteoporotic fractures with age compared to one in five men, and 50-year-old white women have a 16% risk of experiencing an osteoporotic vertebral fracture compared to white men of the same age at 5% (International Osteoporosis Foundation 2023). Bioarchaeologists use this modern clinical data as a foundation for interpreting past bone loss, and need to fully account for the social and cultural factors that constantly influence bone health, morphology, and maintenance (Agarwal and Beauchesne 2011). This weakness is particularly significant in studies of female samples, as there is an assumption that females are inherently more vulnerable and experience greater frailty than males in the same sample (Agarwal and Beauchesne 2011; Beauchesne and Agarwal 2017). This attitude can be summarized neatly in a conclusion drawn by Ericksen (1976): "It appears that a

greater rate of aging bone loss in females is a universal fact of life." As this study and many other biomedical and bioarchaeological studies illustrate, bone loss with age is *not* a universal experience, regardless of sex. Instead, it is a process strongly influenced by social, cultural, and individual factors.

The social and cultural variables in New France likely played an essential role in the results of this study, particularly in the varied results in the youngest age category. While it is possible that individuals in this age group may not have reached peak bone mass yet at the time of death, other studies do not show this same amount of extreme variation before peak bone mass is reached. Significant genetic differences can be ruled out as a contributor to the widely spread results, as the genetic homogeneity of New France is particularly well documented (Scriver 2001). Instead, the vast differences in cortical bone amount in the young sample of Pointe-aux-Trembles and Sainte-Marie can be explored further through the complex social history of New France.

The unusual results in the youngest female sample were likely impacted by the status and expectations of women in Quebec during this era. Women in the French-Canadian colonies were typically married in their late teens or early 20s, and became mothers shortly after marriage. The fertility rate in Lower Canada was higher than that of women in France, and the interval between births was short (Landry 1993). Due to this high fertility and shortened interbirth interval, women in New France likely had a child every two years, from the time of marriage until they reached menopause (Greer 1997). It was not unusual for the women in the French-Canadian colonies to have ten or more children in their lifetime. Although the extent to which pregnancy and lactation result in visible differences in bone amount in archaeological

M.A. Thesis- R. Christenson, McMaster University - Anthropology

skeletons is debated, clinical data has documented that pregnant and breastfeeding women undergo detectable reductions in BMD to meet fetal and neonatal calcium requirements (Kovacs 2016; O'Brien 2021). Pregnancy is clinically associated with a 1-4% decrease in total bone amount across the skeleton, but this amount is recovered within 18-20 months after birth (Ensom et al. 2002; Karlsson et al. 2001; Møller et al. 2012; O'Brien et al. 2021; Pearson et al. 2004). While this reduction in bone amount is not associated with an increase in the risk of osteoporosis, women in New France were undergoing subsequent pregnancies during this recovery period, which would have prevented them from reaching their peak bone mass.

These women also show the highest cortical area amounts, as is expected in the youngest age category. The females in the upper quartile of the 15-29 age category have more cortical bone than the males in the group, which is unusual by modern clinical standards. During the 1700s and 1800s, French-Canadian women were not only responsible for domestic labor, many women performed manual labor alongside their husbands to maintain their farms and livestock (Dumont et al. 1987; Greer 1997). This may have provided the mechanical stimulation necessary to accrue more bone mass.

In context with the social and cultural realities that these women occupied, it is not so unusual that they would represent the highest and lowest bone amounts in the total sample. Further social and cultural factors that may have influenced the results for both sexes will be discussed in Section 5.4.

5.3 Osteobiographical Accounts

5.3.1 Pointe-aux-Trembles

5.3.1.1 7A2-S2

As a middle-aged female (30-49), individual 7A2-S2 shows the highest CAI_{ROI} of the individuals from Pointe-aux-Trembles, and is the second highest in the total combined sample from Pointe-aux-Trembles and Sainte-Marie, calculated at 69.045 (Figure 5.1). In contrast, the micro-CT scan of this individual's rib fragment indicates very little cortical bone, and a sparse trabecular structure, with a mild defect score of 6.5.

The high CAI_{ROI} of this individual suggests they obtained the proper nutrients to achieve a high peak bone mass during skeletal development, and had the ability to maintain this bone mass throughout adulthood. It is likely that this individual had children, experienced pregnancyrelated bone loss, and managed to recover their pre-pregnancy bone amount. Modern clinical studies have observed that BMD can return to pre-pregnancy levels within 12-18 months (Ensom et al. 2002; Karlsson et al. 2001; Møller et al. 2012; O'Brien et al. 2021; Pearson et al. 2004). Fertility decreases in the third decade of a female's life (Committee on Clinical Consensus—Obstetrics 2022); therefore, it is possible that they were no longer able to conceive, or conception became difficult and resulted in a longer inter-birth interval. Assuming that this individual was indeed a mother, based on the fertility data of the French-Canadian colonies, the high CAI_{ROI} value obtained from their metacarpal suggests that they had not been pregnant or breastfeeding for an extended amount of time—long enough that their bone amount was able to recover.



Figure 5.1. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 7A2-S2 (red dot) in the sub-sample (n=10).

5.3.1.2 7A2-S11

Individual 7A2-S11 is a female between the age of 30-49. This individual has a CAI_{ROI} value of 32.956, one of the lowest in the Pointe-aux-Trembles sample and in the lowest 10% of the total combined sample. Initial assessments indicated this individual was possibly suffering from osteomalacia, particularly due to the state of the pelvis and hip joint. A fracture near the left acetabulum shows evidence of healing and extensive new bone formation in the area around the joint. The fracture is bordered by a poorly mineralized callus, a feature which Brickley et al. (2020, pp. 117-120) describe in pseudofractures caused by osteomalacia. The head of the left femur is flattened and articulates with the posterior aspect of the proximal femur, creating a false joint in the fractured acetabulum. The micro-CT analysis of this individual's rib assigned this individual a defect score of 10.5, which was a relatively high compared to others in the sample (Figure 5.2).



Figure 5.2. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 7A2-S11 (red dot) in the sub-sample (n=10).

The skeletal evidence presented here suggests that individual 7A2-S11 consistently experienced poor health during their lifetime. Their low CAI_{ROI} may have been caused by interrupted bone development due to malnutrition or infection during growth, the strain of serial pregnancies and periods of breastfeeding, skeletal resorption as a consequence of disease, or a combination of these, and many more, factors. The state of the hip joint likely limited their mobility and greatly decreased their quality of life further (Schnell et al. 2010). Although this individual is young by modern standards, it is important to note that the settlers of French Canada usually did not live past the age of 40 (Boulet 2021); therefore, despite this individual's experience with potential infectious disease and malnutrition, coupled with the challenges presented by the pathological conditions in their hip, they did manage to reach the average life expectancy of their time. It is likely that individual 7A2-S11 had been unwell for much of their life, and their death was likely not sudden.

5.3.1.3 7A2-S20

There is less skeletal data available for this individual compared to others in the sample, due to the condition of their burial. The site report notes this individual's burial was "very disturbed" and truncated by another trench, leaving only skull fragments, thoracic fragments, and a select few bones of the hand. Using the skull, this individual was estimated as female, and the status of the vertebral fusion indicated they were between the age of 20 and 29, placing them in the youngest age category for this study. The CAI_{ROI} value for this individual is considerably low at 29.382, placing them in the lowest 5% of the total sample (Figure 5.3). The cortical bone of the rib is thin, and the trabeculae are sparse for this age category. It is possible that this individual had undergone some form of nutritional stress in the past that impacted their bone development, but it is unclear if this is related to osteomalacia as the visible defects in the rib scan only had a score of 5.5.

Similar to individual 7A2-S11, it is possible that this individual was unwell for an extended period of time, and this contributed to their death. The low bone amount observed in this individual could be due to complications during skeletal development that impaired skeletal growth, or temporary loss of bone due to pregnancy and breastfeeding, for example.



Figure 5.3. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 7A2-S20 (red dot) in the sub-sample (n=10).

5.3.1.4 7A9-S38

This individual was estimated to be a male between the ages of 14 and 19 based on the fusion status of several epiphyseal sites across the skeleton, making them potentially the youngest individual in the entire sample examined in this study. The present elements of their skeleton bore no obvious signs of pathology, and no mineralization defects were visible in the scan of their rib, which showed an otherwise robust cortical area and a healthy trabecular structure. The CAI_{ROI} value obtained from the second metacarpal is 67.181, the second highest value from Pointe-aux-Trembles and within the top 10% of the total combined sample (Figure 5.4). Based on the skeletal evidence, they accrued a healthy amount of bone for their cohort and likely would have accumulated more bone as their skeleton reached maturity.

It is likely that this individual performed manual labor, either on their family's farm or in Montréal. If they were on the older side of the age range assigned to them, it is also fairly likely
that they were married and had started a family. Based on the available characteristics of this individual, they were probably a stereotypical, French-Canadian laboring male who was in good health. It is quite likely, then, that this individual died suddenly, and it is unclear what factors may have contributed to their death.



Figure 5.4. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 7A9-S38 (red dot) in the sub-sample (n=10).

5.3.1.5 7A9-S46

This individual was located in a densely populated sector of the cemetery, positioned beneath individual 7A2-S20 and similarly truncated by a trench. Much of their skeleton was missing or badly fragmented. It was possible to estimate their sex as female, and the features of the pelvis indicate they are likely of an advanced age: the morphology of the auricular surface indicates they are above the age of 60, while the presence of iliac exostoses suggests they may be above the age of 80 (Milner and Boldsen 2012). Their cortical area amount is 49.569, which falls near the median CAI_{ROI} in both the Pointe-aux-Trembles and total samples (Figure 5.5). This individual received a mineralization defect score of 6.5.



Figure 5.5. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 7A9-S46 (red dot) in the sub-sample (n=10).

The defects in the rib indicate this individual may have experienced osteomalacia. Given that this individual was very old, the onset of osteomalacia may have been caused by the cumulative inefficiency of physiological processes with age (Clegg et al. 2013; Weiss 2011). As individuals age, the ability to regulate homeostasis in the body begins to fail (Clegg et al. 2013). Systems essential to the regulation of vitamin D levels commonly become less efficient in the elderly. The synthesis of vitamin D in the skin becomes less efficient in older individuals (Brickley et al. 2020, pp.81-82), and the decrease in kidney function, which synthesizes active vitamin D, is one of the most common complications of aging (Coresh and Astor 2006).

Although this individual may have been experiencing the negative effects of old age at the time of their death, their CAI_{ROI} amount is very high for an older individual. Considering

their bone mass in advanced age is relatively high compared to others in their age cohort, it may be assumed that their peak bone mass was quite high. This suggests they had access to sufficient nutrition during growth and development, adequate mechanical stimulus, a genetic predisposition to high bone amount, or a combination of these factors. Without more skeletal elements it is difficult to make a complete assessment of their health. However, since life expectancy rarely exceeded 40 years of age in the colonies (Boulet 2021), is probable that this individual lived a long, relatively healthy life in Pointe-aux-Trembles.

5.3.1.6 7A11-S19

Individual 7A11-S19 was located in a cluster with five other individuals (including 7A11-S42), and is missing their skull, several ribs, and both feet. Several of the long bones were severely fragmented. Using the auricular surface of the pelvis, they were estimated to be an older male in their 60s. The CAI_{ROI} value calculated for this individual is 43.313, placing them in the lowest quartile in the Pointe-aux-Trembles sample and ranking them middle-low in the total sample (Figure 5.6). The micro-CT scan of their rib shows the thinning of the cortical bone through increased porosity and the coalescence of these porous areas, and the trabecular structure is almost nonexistent. The visible defects in the rib scan were quite severe, at a score of 12.5.

It is possible that this individual was experiencing age-related bone loss based on the low CAI_{ROI} value of their second metacarpal, the porosity of their rib fragment, and the breakdown of the trabeculae in their rib. Their age, several decades above the average life expectancy, suggests that they had been relatively healthy throughout their lifetime until this

point. The severe mineralization defects, then, may be due to the physiological changes associated with age, explained in Section 5.3.1.5, or may be due to another pathological condition that is not age-related. It is possible that this individual died of old age rather than a form of illness.



Figure 5.6. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 7A11-S19 (red dot) in the sub-sample (n=10).

5.3.1.7 7A11-S42

This individual was located in the same sector as 7A11-S19, and was located beneath three other individuals within this cluster. Many of their skeletal elements are poorly preserved or absent, as they were truncated by a trench and were very disturbed. The features of the skull suggest this individual is female, and the status of vertebral fusion indicates that they were 16-20 years old, one of the youngest in the sample. Due to the incomplete nature of the skeleton and the poor preservation of the remains that were present, there was no gross pathology visible. However, this individual has the second-lowest CAI_{ROI} value obtained in both the Pointeaux-Trembles sample and in the total study sample, calculated at 27.813 (Figure 5.7). Micro-CT scans of the rib show very thin cortical bone and a sparse trabecular structure, and they were assigned a mineralization defect score of 4.5.



Figure 5.7. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 7A11-S42 (red dot) in the sub-sample (n=10).

The limited skeletal information available suggests that this individual was not in good health at the time of their death. Their estimated age at death indicates that they had not completed skeletal growth and therefore did not have the chance to accumulate their peak bone mass. However, the observed CAI_{ROI} value was so low that it is unlikely they would have accumulated enough bone to be comparable to others in their age cohort with high CAI_{ROI} once they reached skeletal maturity. This individual is so young that they could have had only one or two pregnancies before death, ruling out repeated pregnancies and periods of lactation as a source of the low bone amount. Instead, it is likely that malnutrition or infectious disease impeded skeletal growth in this individual, and impacted their quality of life.

5.3.2 Sainte-Marie

5.3.2.1 2B9

Individual 2B9 has the lowest CAI_{ROI} value in the Sainte-Marie group, at 31.012, and is in the lowest 10% of the total sample (Figure 5.8). The scan of their rib shows a robust cortical area, with a low number of trabeculae. Based on the pelvic features, they are most likely a female above the age of 50. The rib scan received a defect score of 1.0, indicating there were no significant defects present, but it is possible that they were experiencing the effects of agerelated bone loss. If this individual had consecutive pregnancies until they reached menopause, they would not have the ability to compensate for lost bone before they reached their peak bone amount, and subsequently began to lose bone due to the uncoupling of bone deposition and bone resorption with age. Despite this, this individual lived beyond the average life expectancy of the time, indicating that they had been relatively healthy for much of their life.



Figure 5.8. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 2B9 (red dot) in the sub-sample (n=10).

5.3.2.2 2E13

Based on the incomplete fusion of the vertebral epiphyses, individual 2E13 is likely between the ages of 16 and 20, and is likely female based on the morphology of their skull. This individual has the highest CAI_{ROI} value of both the Sainte-Marie and the total sample at 73.856 (Figure 5.9). The micro-CT scan of the rib indicates a robust cortical area and consistent trabecular network, which showcases evidence of severe mineralization defects.



Figure 5.9. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 2E13 (red dot) in the sub-sample (n=10).

Although they have a high bone amount, the appearance of their bone in the micro-CT scan of the ribs shows severe defects in the bone microstructures, with the highest overall defect score of 14.5. This individual had osteomalacia when they died, a condition that could be indicative of vitamin D deficiency, or another chronic disease that would have resulted in the malabsorption of the essential nutrients for developing good quality bone. This individual, one of the youngest in the sample, has the highest bone amount from both Pointe-aux-Trembles

and Sainte-Marie, despite never reaching their peak bone mass. These factors suggest this individual did not experience long-term vitamin D deficiency, as they were able to accrue and maintain high levels of bone during growth, and instead developed osteomalacia shortly before their death.

5.3.2.3 2G4

Individual 2G4 was assessed as male within the 15-29 age category, although the assessment of their pelvis results in a more precise estimate of 20-24 years old. The CAI_{ROI} value measured from this individual's second metacarpal is 43.449, which falls in the middle-low range of CAI_{ROI} values for both the Sainte-Marie and the total samples (Figure 5.10).



Figure 5.10. Cortical area index (CAI_{ROI}) and mineralization defect score of Individual 2G4 (red dot) in the sub-sample (n=10).

The ribs show a healthy cortical bone structure, although the trabecular geometry is quite sparse. There were some mineralization defects present, but they were not consistent

throughout the rib and were mild in appearance. This individual was assigned an overall defect score of 2.0, which classifies them as having no defects. Despite the low CAI_{ROI} value, this individual appeared to have a healthy skeleton, indicating that their death may have been sudden.

5.4 Osteoporosis, Vitamin D Deficiency, and Community Health in Quebec

In colonial French-Canada, disease, starvation, and economic fluctuations were linked, and constantly influenced community health. Although communities in the French colonies were somewhat isolated, they were constantly experiencing epidemic events transmitted from a range of sources. Tuberculosis, pneumonia, measles, smallpox, typhus, and cholera were some of the most common infectious diseases experienced in Quebec during the 1700s and 1800s (Boulet 2021; Bruckner et al. 2018; Dawson 1854; Greer 1997; Sheriff 1849). Poor sanitation in urban centers such as Montréal and Québec City, and the spread of disease to the colonial ports from ships were the primary causes of infection. The low population density in the colonies meant urban diseases did not always travel to rural communities such as Pointeaux-Trembles and Sainte-Marie (Bruckner et al. 2018; Charters 2009), but the construction of the Chemin du Roy that connected much of New France could have created a pathway for travelers to spread disease (Desjardins 2010; Robichaud 2012). Medical care in the French-Canadian colonies was primarily given by military surgeons, rather than physicians trained in clinical medicine, and thus their experience in identifying, describing, and treating diseases was limited (Boulet 2021). Rural care was primarily left to these surgeons, as the physicians that were present in French Canada during the 1700s and 1800s typically gathered in Québec City,

where they could treat government officials, rather than in villages and *seigneuries*. Therefore, when epidemics did break out in rural communities, they were rarely equipped with the knowledge to properly treat or mitigate infectious diseases. Infectious disease is the primary reason why the life expectancy in the French-Canadian colonies was so low; therefore, it is likely that infectious disease contributed to the deaths of the majority of these individuals. Repeated and prolonged exposure to infectious diseases has consequences for skeletal development (Section 2.3.4), which could explain some of the low CAI_{ROI} values observed in this sample. The observation of mineralization defects in the majority of the individuals selected for micro-CT analysis suggests the possibility that interrupted vitamin D metabolism was widely experienced in the colonies. This could be due to the insufficient seasonal exposure to UVB experienced at this latitude, but it is also possible that these interruptions occurred as a result of behavioral changes in those experiencing infectious disease. For instance, those who are ill may experience appetite loss and may avoid eating nutritious meals, or may be unable to properly digest and absorb nutrients from nutritive foods due to complications such as vomiting or diarrhea. As well, individuals may stay indoors during their recovery, limiting their exposure to the sun. Avoiding sunlight and nutritious foods as a consequence of infectious disease can result in vitamin D deficiency, causing systemic metabolic disruptions. It is another possibility that the frequency of osteomalacia noted in this study is reflected in individuals who were already ill before death, not those who recovered and survived. This concept is addressed in more detail in Section 5.5.

Starvation and malnutrition were common experiences in the French-Canadian colonies, which has implications for skeletal health. The colonies frequently experienced food shortages,

particularly after the British took possession of New France in 1759 (Ouellet 1980). The blockade of the St. Lawrence River by the British navy during the Seven Years' War created the most severe famine experienced in French Canada during the colonial era (Charters 2009; Marston 2001). When the British gained control of the French-Canadian colonies after the Seven Years' War, changes in the political and economic policies of the colonies often caused hardship for the French colonists (Charters 2009; Ouellet 1980). The British economy absorbed most of the excess capital and resources during times of economic expansion, but high taxes and required export quotas alienated *habitants*, impacting their ability to maintain selfsufficiency on their farms (Ouellet 1980). It is possible that some individuals examined in this study had lived during this tumultuous period. Independently of the stress of war and taxes, starvation during long winters and periods of crop failure resulted in famine in the colonies.

Malnutrition and infectious disease are often linked, as they interact to increase the severity of one another. Malnutrition can negatively impact immune function, not only making individuals more susceptible to disease but making recovery more difficult (Thapar and Sanderson 2004). Infectious disease can result in nutrient loss through the malabsorption and excretion of vitamins or minerals, thus requiring the breakdown of tissues within the body to access sources of stored nutrients (Brown 2003; Guerrant et al. 2008; Thapar and Sanderson 2004). This leads to a cycle of infection and malnutrition that can impact skeletal growth and peak bone mass in juveniles and adolescents, and skeletal maintenance in adults (Norman et al. 2021; Prentice and Bates 1993). Understanding the consequences of malnutrition and infectious disease and placing them in the context of the French-Canadian colonies, which

consistently experienced both, explains why low bone amount and mineralization defects are prevalent in this sample.

There are possible genetic factors that could have contributed to the widespread experience of vitamin D deficiency in Quebec, and while they are important to mention, it is unlikely that these genetic influences impacted the sample of this study. The French-Canadian population of modern Quebec is descended from approximately 8500 original settlers of the colonies in the 1600s, resulting in limited genetic variation in the generations following these initial settlers (Laberge et al. 2005; Scriver 2001). The genetic isolation of the French colonies resulted in a population that experienc an unusually high prevalence of Mendelian diseases. The population of French-Canada currently has the highest prevalence of pseudo-vitamin D deficiency rickets (PDDR) in the world (Glorieux and Pettifor 2014). Genetic linkage studies have estimated that the genetic carrier frequency for PDDR is 1/26 in southern Quebec, compared to a frequency of <1/500 elsewhere (Labuda et al. 1996). PDDR is an autosomal recessive disorder characterized by a mutation in the genetic coding of the enzymes responsible for the hydroxylation of vitamin D into its usable form (Scriver 2001; Glorieux and Pettifor 2014; Laberge et al. 2005). It is unlikely that individuals with PDDR can survive until adulthood without medical intervention (Glorieux and Pettifor 2014), ruling this out as a factor contributing to the health of adults in the past. However, this adds another avenue of exploration in further studies of infant mortality in these communities.

Ultimately, the social, cultural, and environmental contributors to the low bone amounts noted in this study are likely more influential than age-related processes in this sample. Individuals in the French-Canadian colonies rarely lived beyond the age of 40, and aging

beyond the sixth decade of life, the period where age-related bone loss typically becomes classified as osteoporosis, was even more uncommon (Boulet 2021). The presence of at least mild mineralization defects in nearly all of the scanned rib samples suggests that vitamin D deficiency in this sample was systemic, caused by community-wide health events, genetic susceptibility to vitamin D deficiency, or both.

5.5 Limitations and Future Research

This study operates on the assumption that the sample represents the health status of the entire communities at Pointe-aux-Trembles and Sainte-Marie. Realistically, the sample size of this study is too small to be representative of the entire population. The availability of viable skeletal elements was limited due to taphonomic damage, the removal of some individuals periodically as the cemeteries were expanded, the inability to excavate the remains of all individuals at the sites, and the commingling of remains due to the dense burials. Limitations in time and resources only allowed 10 rib samples to be used in the micro-CT analysis, which limits the scope of our interpretation of vitamin D deficiency and osteomalacia at the population level. The visibility of mineralization defects in several of the rib samples was obscured by damage caused by fungal tunnelling (Jans et al. 2004; Mazzoli et al. 2018).

The concept of the osteological paradox, explored by Wood et al. (1992) and revisited by DeWitte and Stojanowski (2015), outlines some of the major challenges and limitations of paleopathological research. While these limitations do not apply to all paleopathological studies (DeWitte and Stojanowski 2015), they do apply to this research. For instance, the

concept of selective mortality suggests that assemblages only represent individuals who died, not those who were at risk of death and survived. We cannot observe whether individuals experienced low bone amount during childhood due to illness or malnutrition and subsequently recovered when the stress event ended, for example. This study also assumes that individuals in a population are equal in their susceptibility to disease and risk of death. Ultimately, individuals in a population vary in their frailty due to a multitude of genetic, socioeconomic, environmental, and temporal factors.

Despite these limitations, this research has implications for future studies on skeletal health. This study is the first to implement a quantitative system for recording mineralization defects in micro-CT scans, allowing for the consistent collection of data that can be used for inter- and intrapopulation comparisons. Future studies with the resources to perform full pathological analyses of the skeletons and examine more than a small subset of the sample to view mineralization defects are recommended. Although it is a time-consuming process, such detailed studies can provide even deeper insights into past health. Further explorations of metabolic bone disease within other historical French-Canadian communities will create a wider understanding of health and disease in Quebec during the colonial era. Comparing and contrasting this data with European assemblages can provide further information on the ways different cultures, societies, and environments influenced biological processes in past populations.

Chapter 6: Conclusion

Metabolic bone diseases such as osteoporosis and osteomalacia are heavily influenced by cultural and environmental factors, in addition to biological factors. The aim of this thesis was to determine patterns of osteoporotic bone loss and osteomalacia in historical French-Canadian communities to provide insight into lived experiences and community health in the past. This was completed through the application of a newly developed method of metacarpal radiogrammetry (Gilmour et al. 2021) and a pilot project for interpreting mineralization defects developed for this thesis.

This research created an opportunity to apply new and developing analytical techniques to fragmentary archaeological remains, allowing researchers to gather information beyond the parameters of traditional data collection techniques. The remains recovered from Pointe-aux-Trembles and Sainte-Marie were generally poorly preserved; therefore, it was necessary to use techniques that did not rely on complete elements for analysis. Using the metacarpal radiogrammetry method developed by Gilmour et al. (2021), it was possible to use fragmentary metacarpals that would have otherwise been excluded in traditional radiogrammetric methods. The technique developed for scoring mineralization defects in the ribs requires only a 1cm area of bone for analysis, allowing for the inclusion of fragmentary ribs.

Medical research emphasizes the importance of biological determinants of metabolic bone disease, particularly genetics, age, and sex. It was expected to view a pattern of bone accumulation and loss that follows the age and sex trends established by clinical data, as they have been observed in many bioarchaeological studies of age-related bone loss (see Curate et

al. 2013, Ericksen 1976, Glencross and Agarwal 2011, Van Gerven et al. 1969). By exploring patterns of bone amount by sex and age in Pointe-aux-Trembles and Sainte-Marie through metacarpal radiogrammetry, it was determined that bone amount was not significantly linked to age or sex. Instead, the youngest individuals in the sample experienced the highest bone amounts, as expected, and the lowest bone amount, an unexpected result. Females in the youngest age category show particularly variable bone amounts, indicating that bone amount is not strictly biologically determined.

Seven of the ten individuals analyzed for osteomalacia showed at least some evidence of mineralization defects throughout their ribs. It was expected that evidence of osteomalacia would be related to osteoporosis; however, there were no patterns in the presence or severity of defects by sex, age, or cortical area index (CAI_{ROI}), suggesting that vitamin D deficiency may have been a widespread issue in Pointe-aux-Trembles and Sainte-Marie.

The results of this thesis were interpreted through a historical and cultural lens, providing insight into individual and community health in French Canada in the 1700s and 1800s. Through this approach, social determinants of health such as social roles, food insecurity, sanitation, economic fluctuations, and the environment were given as much analytical weight as the biological factors of bone health.

Within the cultural and social contexts explored through biocultural theory, a lived experience framework made it possible to further understand these individuals not as just "constituents of samples" (Hosek and Robb 2019), but as dynamic humans. Approaching lived experience through osteobiography allowed a deeper exploration of the co-occurrence of

osteoporosis and osteomalacia on the individual level, and in understanding individual experiences, it was possible to extrapolate this relationship to the total sample level.

Ultimately, this study suggests that health patterns observed in Pointe-aux-Trembles and Sainte-Marie may deviate from health patterns reported in modern clinical literature and other bioarchaeological studies. The ways society and culture developed in the St. Lawrence region created communities that were distinct from other North American colonies and European cities. Patterns of bone amount that have not been observed in other archaeological samples, and the prevalence of mineralization defects in the sample implies that individuals in these communities may have experienced widespread health issues that impacted their ability to accumulate and maintain bone during their lifetime. This claim is supported by historical sources, which document frequent periods of illness and starvation.

Although the sample size in this study is too small to make unequivocal statements regarding the health of these communities on the population level, the osteobiographical approach provides a unique view inside the lived experiences of these individuals that is otherwise invisible from the sample level. Further explorations of the interaction between disease, malnutrition, growth, and aging can create further insight into the lived experiences in these two communities.

References

- Agarwal, S. C. (2012). The Past of Sex, Gender, and Health: Bioarchaeology of the Aging Skeleton. *American Anthropologist*, *114*(2), 322–335. <u>https://doi.org/10.1111/j.1548-1433.2012.01428.x</u>
- Agarwal, S. C. (2019). Understanding Bone Aging, Loss, and Osteoporosis in the Past. In M. A. Katzenberg & A. L. Grauer (Eds.), *Biological Anthropology of the Human Skeleton* (Third Edition, pp. 385–414). John Wiley & Sons, Inc.
- Agarwal, S. C. (2021). What is normal bone health? A bioarchaeological perspective on meaningful measures and interpretations of bone strength, loss, and aging. *American Journal of Human Biology*, *33*(5), e23647. <u>https://doi.org/10.1002/ajhb.23647</u>
- Agarwal, S. C., & Beauchesne, P. (2011). It is not Carved in Bone. In *Social Bioarchaeology* (pp. 312–332). John Wiley & Sons, Ltd. <u>https://doi.org/10.1002/9781444390537.ch11</u>
- Agarwal, S. C., & Stuart-Macadam, P. (2003). An Evolutionary and Biocultural Approach to Understanding the Effects of Reproductive Factors on the Female Skeleton. In S. C. Agarwal & S. D. Stout (Eds.), *Bone Loss and Osteoporosis: An Anthropological Perspective*. Springer Science+Business Media.
- Albert, M. A., & Maier, C. A. (2013). Epiphyseal Union of the Cervical Vertebral Centra: Its Relationship to Skeletal Age and Maturation of Thoracic Vertebral Centra. *Journal of Forensic Sciences*, *58*(6), 1568–1574. <u>https://doi.org/10.1111/1556-4029.12232</u>
- Albert, M., Mulhern, D., Torpey, M. A., & Boone, E. (2010). Age Estimation Using Thoracic and First Two Lumbar Vertebral Ring Epiphyseal Union. *Journal of Forensic Sciences*, 55(2), 287–294. <u>https://doi.org/10.1111/j.1556-4029.2009.01307.x</u>
- Almeida, M., Laurent, M. R., Dubois, V., Claessens, F., O'Brien, C. A., Bouillon, R., Vanderschueren, D., & Manolagas, S. C. (2017). Estrogens and Androgens in Skeletal Physiology and Pathophysiology. *Physiological Reviews*, *97*(1), 135–187. <u>https://doi.org/10.1152/physrev.00033.2015</u>
- Arnold, J. S. (1973). Amount and quality of trabecular bone in osteoporotic vertebral fractures. *Clinics in Endocrinology and Metabolism*, 2(2), 221–238. <u>https://doi.org/10.1016/S0300-595X(73)80041-6</u>
- Barnett, E., & Nordin, B. E. C. (1960). The Radiological Diagnosis of Osteoporosis: A New Approach. *Clinical Radiology*, *11*(3), 166–174.

- Bauman, W., & Cardozo, C. (2013). Immobilization osteoporosis. In R. Marcus, D. Feldman, D.
 Dempster, M. Luckey, & J. Cauley (Eds.), *Osteoporosis* (Fourth Edition, pp. 1139–1171). Elsevier Science & Technology.
- Beauchesne, P., & Agarwal, S. C. (2014). Age-Related Cortical Bone Maintenance and Loss in an Imperial Roman Population. *International Journal of Osteoarchaeology*, 24(1), 15– 30. <u>https://doi.org/10.1002/oa.1303</u>
- Beauchesne, P., & Agarwal, S. C. (2017). A multi-method assessment of bone maintenance and loss in an Imperial Roman population: Implications for future studies of age-related bone loss in the past. American Journal of Physical Anthropology, 164(1), 41–61. <u>https://doi.org/10.1002/ajpa.23256</u>
- Bhan, A., Qiu, S., & Rao, S. D. (2018). Bone histomorphometry in the evaluation of osteomalacia. *Bone Reports*, *8*, 125–134. <u>https://doi.org/10.1016/j.bonr.2018.03.005</u>
- Bischoff-Ferrari, H. A., Giovannucci, E., Willett, W. C., Dietrich, T., & Dawson-Hughes, B. (2006). Estimation of optimal serum concentrations of 25-hydroxyvitamin D for multiple health outcomes. *The American Journal of Clinical Nutrition*, 84(1), 18–28. <u>https://doi.org/10.1093/ajcn/84.1.18</u>
- Black, D. M., & Rosen, C. J. (2016). Postmenopausal Osteoporosis. *New England Journal of Medicine*, 374(3), 254–262. <u>https://doi.org/10.1056/NEJMcp1513724</u>
- Blake, G. M., & Fogelman, I. (2007). The role of DXA bone density scans in the diagnosis and treatment of osteoporosis. *Postgraduate Medical Journal*, *83*(982), 509–517. <u>https://doi.org/10.1136/pgmj.2007.057505</u>
- Börjesson, A. E., Lagerquist, M. K., Liu, C., Shao, R., Windahl, S. H., Karlsson, C., Sjögren, K., Movérare-Skrtic, S., Antal, M. C., Krust, A., Mohan, S., Chambon, P., Sävendahl, L., & Ohlsson, C. (2010). The role of estrogen receptor α in growth plate cartilage for longitudinal bone growth. *Journal of Bone and Mineral Research*, *25*(12), 2690– 2700. https://doi.org/10.1002/jbmr.156
- Boskey, A. L. (2010). The Biochemistry of Bone: Composition and Organization. In E. S. Orwoll, J.
 P. Bilezikian, & D. Vanderschueren (Eds.), Osteoporosis in Men: The Effects of Gender on Skeletal Health (Second Edition). Elsevier Academic Press.
- Boulet, L.-P. (2021). Respiratory health in Canada before 1800. *Canadian Journal of Respiratory, Critical Care, and Sleep Medicine, 5*(5), 334–340. <u>https://doi.org/10.1080/24745332.2018.1531685</u>

- Boylorn, R. M. (2008). Lived Experience. In L. M. Given (Ed.), *The SAGE Encyclopedia of Qualitative Research Methods* (Vols. 1 & 2). SAGE Publications Inc.
- Brandão, J. A. (1997). "Your Fyre Shall Burn No More:" Iroquois Policy Toward New France and Its Native Allies to 1701. University of Nebraska Press.
- Brickley, M. B., & Buckberry, J. (2015). Picking up the pieces: Utilizing the diagnostic potential of poorly preserved remains. *International Journal of Paleopathology*, 8, 51–54. <u>https://doi.org/10.1016/j.ijpp.2014.08.003</u>
- Brickley, M. B., Ives, R., & Mays, S. (2020). *The Bioarchaeology of Metabolic Bone Disease* (Second Edition). Elsevier Academic Press.
- Brickley, M., Mays, S., & Ives, R. (2007). An investigation of skeletal indicators of vitamin D deficiency in adults: Effective markers for interpreting past living conditions and pollution levels in 18th and 19th century Birmingham, England. *American Journal of Physical Anthropology*, 132(1), 67–79. https://doi.org/10.1002/ajpa.20491
- Brooks, S., & Suchey, J. M. (1990). Skeletal age determination based on the os pubis: A comparison of the Acsádi-Nemeskéri and Suchey-Brooks methods. *Human Evolution*, 5(3), 227–238. <u>https://doi.org/10.1007/bf02437238</u>
- Brown, K. H. (2003). Diarrhea and Malnutrition. *The Journal of Nutrition*, *133*(1), 328S-332S. https://doi.org/10.1093/jn/133.1.328S
- Bruckner, T. A., Gailey, S., Hallman, S., Amorevieta-Gentil, M., Dillon, L., & Gagnon, A. (2018). Epidemic cycles and environmental pressure in colonial Quebec. *American Journal* of Human Biology, 30(5), e23155. <u>https://doi.org/10.1002/ajhb.23155</u>
- Buikstra J, Ubelaker D. 1994. Standards for Data Collection from Human Skeletal Remains: Proceedings of a Seminar at the Field Museum of Natural History (Arkansas archaeology research series 44). Fayetteville Arkansas Archaeological Survey.
- Chaplin, G., & Jablonski, N. G. (2013). The Human Environment and the Vitamin D Compromise: Scotland as a Case Study in Human Biocultural Adaptation and Disease Susceptibility. *Human Biology*, 85(4), 529–552. <u>https://doi.org/10.3378/027.085.0402</u>
- Charters, E. M. (2009). Disease, Wilderness Warfare, and Imperial Relations: The Battle for Quebec, 1759—1760. *War in History*, *16*(1), 1–24. <u>https://doi.org/10.1177/0968344508097615</u>

- Choi, M. H., Yang, J. H., Seo, J. S., Kim, Y., & Kang, S.-W. (2021). Prevalence and diagnosis experience of osteoporosis in postmenopausal women over 50: Focusing on socioeconomic factors. *PLOS ONE*, *16*(3), e0248020. <u>https://doi.org/10.1371/journal.pone.0248020</u>
- Cianferotti, L., Gomes, A. R., Fabbri, S., Tanini, A., & Brandi, M. L. (2015). The calcium-sensing receptor in bone metabolism: From bench to bedside and back. *Osteoporosis International*, *26*(8), 2055–2071. <u>https://doi.org/10.1007/s00198-015-3203-1</u>
- Clegg, A., Young, J., Iliffe, S., Rikkert, M. O., & Rockwood, K. (2013). Frailty in elderly people. *The Lancet*, *381*(9868), 752–762. <u>https://doi.org/10.1016/S0140-6736(12)62167-9</u>
- Committee on Clinical Consensus--Obstetrics. (2022). Pregnancy at Age 35 Years or Older. Obstetrics and Gynecology, 140(2), 348–366.
- Compston, J. E., McClung, M. R., & Leslie, W. D. (2019). Osteoporosis. *The Lancet, 393*(10169), 364–376. <u>http://dx.doi.org/10.1016/S0140-6736(18)32112-3</u>
- Coresh, J., & Astor, B. (2006). Decreased Kidney Function in the Elderly: Clinical and Preclinical, Neither Benign. *Annals of Internal Medicine*, *145*(4), 299. <u>https://doi.org/10.7326/0003-4819-145-4-200608150-00010</u>
- Cremers, S., Meier, C., & Seibel, M. J. (2010). Markers of Bone Remodeling and the Aging Skeleton. In E. S. Orwoll, J. P. Bilezikian, & D. Vanderschueren (Eds.), *Osteoporosis in Men: The Effects of Gender on Skeletal Health* (Second Edition, pp. 179–184). Elsevier Academic Press.
- Curate, F., Albuquerque, A., Correia, J., Ferreira, I., de Lima, J. P., & Cunha, E. M. (2013). A glimpse from the past: Osteoporosis and osteoporotic fractures in a Portuguese identified skeletal sample. *Acta Reumatologica Portuguesa*, *38*, 20–27.
- Dahlberg, H. (2021). Open and Reflective Research: Methodological reflections on the importance of body awareness in qualitative research. In J. F. Tantia (Ed.), *The Art and Science of Embodied Research Design: Concepts, Methods and Cases* (pp. 19– 28). Routledge, Taylor & Francis Group.
- Dakhale, G., Hiware, S., Mahatme, M., & Shinde, A. (2012). Basic biostatistics for post-graduate students. *Indian Journal of Pharmacology*, 44(4), 435. <u>https://doi.org/10.4103/0253-</u> <u>7613.99297</u>

- Dawson, B. (1854). Practical Views on Cholera, and on the Sanitary, Preventative and Curative Measures to be Adopted in the Event of a Visitation of the Epidemic.
- de Boer, H. H. (Hans), & Van der Merwe, A. E. (Lida). (2016). Diagnostic dry bone histology in human paleopathology. *Clinical Anatomy*, 29(7), 831–843. <u>https://doi.org/10.1002/ca.22753</u>
- Dechêne, L. (1992). *Habitants and Merchants in Seventeenth Century Montreal* (L. Vardi, Trans.). McGill-Queen's University Press.
- Desjardins, P. (2010). « Du chemin qui marche » au chemin du Roy à la Pointe-aux-Trembles. *Histoire Québec*, *16*(2), 11–14.
- Dewey, J. R., Armelagos, G. J., & Bartley, M. H. (1969). Femoral Cortical Involution in Three Nubian Archaeological Populations. *Human Biology*, *41*(1), 13–28.
- DeWitte, S. N., & Stojanowski, C. M. (2015). The Osteological Paradox 20 Years Later: Past Perspectives, Future Directions. *Journal of Archaeological Research*, 23(4), 397–450. <u>https://doi.org/10.1007/s10814-015-9084-1</u>
- Dodgson, J. E. (2023). Phenomenology: Researching the Lived Experience. *Journal of Human Lactation*, *39*(3), 385–396. <u>https://doi.org/10.1177/08903344231176453</u>
- Doran, P. M., & Khosla, S. (2004). Senile Osteoporosis. In J. E. Henderson & D. Goltzman (Eds.), *The Osteoporosis Primer* (pp. 225–235). Cambridge University Press.
- Dumont, M., Jean, M., Lavigne, M., & Stoddart, J. (1987). *Quebec Women: A History* (R. Gannon & R. Gill, Trans.). The Women's Press.
- Duncan, E. L., & Brown, M. A. (2008). Genetic studies in osteoporosis the end of the beginning. *Arthritis Research & Therapy*, *10*(5), 214. <u>https://doi.org/10.1186/ar2479</u>
- Ekenman, I., Eriksson, S. A. V., & Lindgren, J. U. (1995). Bone density in medieval skeletons. *Calcified Tissue International*, *56*(5), 355–358. <u>https://doi.org/10.1007/BF00301601</u>
- Ellis, C., & Flaherty, M. G. (1992). *Investigating Subjectivity: Research on Lived Experience*. Sage Publications Inc.
- Ensom, M. H. H., Liu, P. Y., & Stephenson, M. D. (2002). Effect of Pregnancy on Bone Mineral Density in Healthy Women: *Obstetrical and Gynecological Survey*, *57*(2), 99–111. <u>https://doi.org/10.1097/00006254-200202000-00022</u>

- Ericksen, M. F. (1976). Cortical bone loss with age in three Native American populations. *American Journal of Physical Anthropology*, 45(3), 443–452. <u>https://doi.org/10.1002/ajpa.1330450306</u>
- Ethnoscop (2006). Site du premier cimetière de Sainte-Marie, CcEs-1 Inventaire etfouilles archéologiques 2003-2004. 12
- Ethnoscop (2016). Interventions archéologiques dans le cadre du projet de construction de la Maison du citoyen à Pointe-aux-Trembles, 2014, BjFi-17.
- Fay, I. (2006). Text, Space, and the Evidence of Human Remains in English Late Medieval and Tudor Disease Culture: Some Problems and Possibilities. In R. Gowland & C. Knüsel (Eds.), Social Archaeology of Funerary Remains (pp. 190–208). Oxbow Books.
- Finlay, L. (2009). Exploring lived experience: Principles and practice of phenomenological research. International Journal of Therapy and Rehabilitation, 16(9), 474–481. <u>https://doi.org/10.12968/ijtr.2009.16.9.43765</u>
- Fraher, L. J., & Watson, P. H. (2004). Local regulators of bone turnover. In J. E. Henderson & D. Goltzman (Eds.), *The Osteoporosis Primer* (pp. 64–72). Cambridge University Press.
- Frost, H. M. (2003). On Changing Views about Age-Related Bone Loss. In S. C. Agarwal & S. D.
 Stout (Eds.), *Bone Loss and Osteoporosis: An Anthropological Perspective* (pp. 19–31). Springer Science+Business Media.
- Gascon-Barré, M. (2004). Vitamin D metabolism. In J. E. Henderson & D. Goltzman (Eds.), *The* Osteoporosis Primer (pp. 88–101). Cambridge University Press.
- Geloso, V., Kufenko, V., & Arsenault-Morin, A. P. (2023). The lesser shades of labor coercion: The impact of seigneurial tenure in nineteenth-century Quebec. *Journal of Development Economics*, *163*, 103091. <u>https://doi.org/10.1016/j.jdeveco.2023.103091</u>
- Giguère, Y., & Rousseau, F. (2000). The genetics of osteoporosis: 'Complexities and difficulties.' *Clinical Genetics*, 57(3), 161–169. <u>https://doi.org/10.1034/j.1399-</u> <u>0004.2000.570301.x</u>
- Gilmour, R. J., Brickley, M. B., Hoogland, M., Jurriaans, E., Mays, S., & Prowse, T. L. (2021). Quantifying cortical bone in fragmentary archeological second metacarpals. *American Journal of Physical Anthropology*, 174(4), 812–821. <u>https://doi.org/10.1002/ajpa.24248</u>

- Glencross, B., & Agarwal, S. C. (2011). An investigation of cortical bone loss and fracture patterns in the neolithic community of Çatalhöyük, Turkey using metacarpal radiogrammetry. *Journal of Archaeological Science*, 38(3), 513–521. <u>https://doi.org/10.1016/j.jas.2010.10.004</u>
- Glorieux, F. H., & Pettifor, J. M. (2014). Vitamin D/dietary calcium deficiency rickets and pseudo-vitamin D deficiency rickets. *BoneKEy Reports*, *3*, 524. <u>https://doi.org/10.1038/bonekey.2014.19</u>
- Gocha, T. P., Robling, A. G., & Stout, S. D. (2019). Histomorphometry of Human Cortical Bone: Applications to Age Estimation. In M. A. Katzenberg & A. L. Grauer (Eds.), *Biological Anthropology of the Human Skeleton* (Third Edition, pp. 145–188). John Wiley & Sons, Inc.
- Goodman, A. H., & Leatherman, T. L. (1998). *Building a New Biocultural Synthesis: Political-Economic Perspectives on Human Biology*. University of Michigan Press. <u>http://ebookcentral.proquest.com/lib/mcmu/detail.action?docID=3414777</u>
- Google. (2023a). [Google Maps image of Pointe-aux-Trembles]. [Map].
- Google. (2023b). [Google Maps image of Sainte-Marie]. [Map].
- Gough Courtney, M., Roberts, J., & Godde, K. (2023). Structural Inequity and Socioeconomic Status Link to Osteoporosis Diagnosis in a Population-Based Cohort of Middle-Older-Age Americans. *INQUIRY: The Journal of Health Care Organization, Provision, and Financing*, 60, 1–14. https://doi.org/10.1177/00469580231155719
- Greer, A. (1997). *The People of New France* (C. Heron & F. Iacovetta, Eds.). University of Toronto Press.
- Grynpas, M. D. (2003). The Role of Bone Quality on Bone Loss and Fragility. In S. C. Agarwal & S.
 D. Stout (Eds.), *Bone Loss and Osteoporosis: An Anthropological Perspective* (pp. 33–46). Springer Science+Business Media.
- Guerrant, R. L., Oriá, R. B., Moore, S. R., Oriá, M. O., & Lima, A. A. (2008). Malnutrition as an enteric infectious disease with long-term effects on child development. *Nutrition Reviews*, 66(9), 487–505. <u>https://doi.org/10.1111/j.1753-4887.2008.00082.x</u>
- Guerri, S., Mercatelli, D., Aparisi Gómez, M. P., Napoli, A., Battista, G., Guglielmi, G., & Bazzocchi, A. (2018). Quantitative imaging techniques for the assessment of osteoporosis and sarcopenia. *Quantitative Imaging in Medicine and Surgery*, 8(1), 60–85. https://doi.org/10.21037/qims.2018.01.05

- Harada, S., & Rodan, G. A. (2003). Control of osteoblast function and regulation of bone mass. *Nature*, 423(6937), 349–355. <u>https://doi.org/10.1038/nature01660</u>
- Hendy, G. N. (2004). The PTH/PTHrP system and calcium homeostasis. In J. E. Henderson & D. Goltzman (Eds.), *The Osteoporosis Primer* (pp. 73–87). Cambridge University Press.
- Hernandez, C. J., Beaupré, G. S., & Carter, D. R. (2003). A theoretical analysis of the relative influences of peak BMD, age-related bone loss and menopause on the development of osteoporosis. Osteoporosis International, 14(10), 843–847. <u>https://doi.org/10.1007/s00198-003-1454-8</u>
- Holck, P. (2007). Bone mineral densities in the prehistoric, Viking-Age and medieval populations of Norway. *International Journal of Osteoarchaeology*, *17*(2), 199–206. https://doi.org/10.1002/oa.870
- Holick, M. F. (2007). Vitamin D Deficiency. New England Journal of Medicine, 357, 266–281.
- Hosek, L., & Robb, J. (2019). Osteobiography: A Platform for Bioarchaeology Research. Bioarchaeology International, 3(1), Article 1. <u>https://doi.org/10.5744/bi.2019.1005</u>
- Huotari, A., & Herzig, K.-H. (2008). Vitamin D and living in northern latitudes—An endemic risk area for vitamin D deficiency. *International Journal of Circumpolar Health*, 67(2–3), 164–178. <u>https://doi.org/10.3402/ijch.v67i2-3.18258</u>
- International Osteoporosis Foundation. (2023). *Epidemiology of osteoporosis and fragility fractures*. International Osteoporosis Foundation. <u>https://www.osteoporosis.foundation/facts-statistics/epidemiology-of-</u> <u>osteoporosis-and-fragility-fractures</u>
- İşcan, M. Y., Loth, S. R., & Wright, R. K. (1984). Metamorphosis at the sternal rib end: A new method to estimate age at death in white males. *American Journal of Physical Anthropology*, *65*(2), 147–156. <u>https://doi.org/10.1002/ajpa.1330650206</u>
- Isojima, T., & Sims, N. A. (2021). Cortical bone development, maintenance and porosity: Genetic alterations in humans and mice influencing chondrocytes, osteoclasts, osteoblasts and osteocytes. *Cellular and Molecular Life Sciences*, 78(15), 5755–5773. <u>https://doi.org/10.1007/s00018-021-03884-w</u>

- Ives, R., & Brickley, M. B. (2004). A procedural guide to metacarpal radiogrammetry in archaeology. International Journal of Osteoarchaeology, 14(1), 7–17. <u>https://doi.org/10.1002/oa.709</u>
- Ives, R., & Brickley, M. (2005). Metacarpal radiogrammetry: A useful indicator of bone loss throughout the skeleton? *Journal of Archaeological Science*, 32(10), 1552–1559. https://doi.org/10.1016/j.jas.2005.04.008
- Jans, M. M. E., Nielsen-Marsh, C. M., Smith, C. I., Collins, M. J., & Kars, H. (2004). Characterisation of microbial attack on archaeological bone. *Journal of Archaeological Science*, *31*(1), 87–95. <u>https://doi.org/10.1016/j.jas.2003.07.007</u>
- Jones, G. (2018). The discovery and synthesis of the nutritional factor vitamin D. *International Journal of Paleopathology*, 23, 96–99. <u>https://doi.org/10.1016/j.ijpp.2018.01.002</u>
- Jørkov, M. L. S., Heinemeier, J., & Lynnerup, N. (2009). The petrous bone-A new sampling site for identifying early dietary patterns in stable isotopic studies. *American Journal of Physical Anthropology*, *138*(2), 199–209. <u>https://doi.org/10.1002/ajpa.20919</u>
- Karlsson, C., Obrant, K. J., & Karlsson, M. (2001). Pregnancy and Lactation Confer Reversible Bone Loss in Humans. Osteoporosis International, 12(10), 828–834. <u>https://doi.org/10.1007/s001980170033</u>
- Karsdal, M. A., Byrjalsen, I., Leeming, D. J., Delmas, P. D., & Christiansen, C. (2008). The effects of oral calcitonin on bone collagen maturation: Implications for bone turnover and quality. Osteoporosis International, 19(9), 1355–1361. <u>https://doi.org/10.1007/s00198-008-0603-5</u>
- Keaveny, T. M., & Yeh, O. C. (2002). Architecture and trabecular bone Toward an improved understanding of the biomechanical effects of age, sex and osteoporosis. *Journal of Musculoskeletal and Neuronal Interactions*, 2(3), 205–208.
- Knudson, K. J., & Stojanowski, C. M. (2008). New Directions in Bioarchaeology: Recent Contributions to the Study of Human Social Identities. *Journal of Archaeological Research*, 16(4), 397–432. <u>https://doi.org/10.1007/s10814-008-9024-4</u>

- Kostenuik, P. (2013). On the Evolution and Contemporary Roles of Bone Remodeling. In R.
 Marcus, D. Feldman, D. Dempster, M. Luckey, & J. Cauley (Eds.), *Osteoporosis* (Fourth Edition, pp. 873–914). Elsevier Science & Technology.
- Kousteni, S. (2010). The Molecular Biology of Sex Steroids in Bone: Similarities and Differences Among the Sexes. In E. S. Orwoll, J. P. Bilezikian, & D. Vanderschueren (Eds.), Osteoporosis in Men: The Effects of Gender on Skeletal Health (Second Edition, pp. 269–281). Elsevier Academic Press.
- Laberge, A.-M., Michaud, J., Richter, A., Lemyre, E., Lambert, M., Brais, B., & Mitchell, G. (2005). Population history and its impact on medical genetics in Quebec: Impact of population history in Quebec. *Clinical Genetics*, 68(4), 287–301. <u>https://doi.org/10.1111/j.1399-0004.2005.00497.x</u>
- Labuda, M., Labuda, D., Cole, D. E. C., Zivtkiewicz, E., Popowska, E., Pronicka, E., Root, A. W., & Glorieux, F. H. (1996). Linkage Disequilibrium Analysis in Young Populations: Pseudo-Vitamin D-Deficiency Rickets and the Founder Effect in French Canadians. *American Journal of Human Genetics*, *59*, 633–643.
- Landry, Y. (1993). Fertility in France and New France: The Distinguishing Characteristics of Canadian Behavior in the Seventeenth and Eighteenth Centuries. *Social Science History*, *17*(4), 577–592.
- Lanske, B., Amling, M., Neff, L., Guiducci, J., Baron, R., & Kronenberg, H. M. (1999). Ablation of the PTHrP gene or the PTH/PTHrP receptor gene leads to distinct abnormalities in bone development. *Journal of Clinical Investigation*, *104*(4), 399–407.
- Laurent, M., Antonio, L., Sinnesael, M., Dubois, V., Gielen, E., Classens, F., & Vanderschueren, D. (2014). Androgens and estrogens in skeletal sexual dimorphism. *Asian Journal of Andrology*, 16(2), 213–222. <u>https://doi.org/10.4103/1008-682X.122356</u>
- Le Moine, J. M. (1876). *Quebec Past and Present: A History of Quebec 1608-1876*. Augustin Cote Publisher.
- Lips, P. (2001). Vitamin D Deficiency and Secondary Hyperparathy- roidism in the Elderly: Consequences for Bone Loss and Fractures and Therapeutic Implications. *Endocrine Reviews*, 22(4), 477–501.
- Lu, Z., Chen, T. C., Zhang, A., Persons, K. S., Kohn, N., Berkowitz, R., Martinello, S., & Holick, M. F. (2007). An evaluation of the vitamin D3 content in fish: Is the vitamin D content adequate to satisfy the dietary requirement for vitamin D? *The Journal of Steroid Biochemistry and Molecular Biology*, *103*(3–5), 642–644. https://doi.org/10.1016/j.jsbmb.2006.12.010

Lucas, C. P. (1901). History of Canada Part 1: New France. Clarendon Press.

- Malabanan, A., Veronikis, I. E., & Holick, M. F. (1998). Redefining vitamin D insufficiency. *The Lancet*, *351*, 805–806.
- Manolagas, S. C. (2000). Birth and Death of Bone Cells: Basic Regulatory Mechanisms and Implications for the Pathogenesis and Treatment of Osteoporosis. *Endocrine Reviews*, *21*(2), 115–137.
- Marston, D. (2001). The Seven Years' War (First Edition). Routledge.
- Matkovic, V., & Landoll, J. D. (2004). Non-genetic determinants of peak bone mass. In J. E. Henderson & D. Goltzman (Eds.), *The Osteoporosis Primer* (pp. 147–169). Cambridge University Press.
- Mays, S. (2018). How should we diagnose disease in palaeopathology? Some epistemological considerations. *International Journal of Paleopathology*, 20, 12–19. <u>https://doi.org/10.1016/j.ijpp.2017.10.006</u>
- Mazzoli, R., Giuffrida, M. G., & Pessione, E. (2018). Back to the past: "Find the guilty bug microorganisms involved in the biodeterioration of archeological and historical artifacts." *Applied Microbiology and Biotechnology*, *102*(15), 6393–6407. <u>https://doi.org/10.1007/s00253-018-9113-3</u>
- Meema, H. E., & Meema, S. (1987). Postmenopausal Osteoporosis: Simple Screening Method for Diagnosis before Structural Failure. *Radiology*, *164*(2), 405–410.
- Meindl, R. S. & Lovejoy, C. O. (1989). Age Changes in the Pelvis: Implications for
 Paleodemography. In M. Y. Iscan (Ed.), Age Markers in the Human Skeleton (pp. 137-168). Charles C. Thomas Publishing.
- Milner, G. R., & Boldsen, J. L. (2012). Transition Analysis: A Validation Study with Known-Age Modern American Skeletons. *American Journal of Physical Anthropology*, 148, 98– 110.
- Minisola, S., Colangelo, L., Pepe, J., Diacinti, D., Cipriani, C., & Rao, S. D. (2021). Osteomalacia and Vitamin D Status: A Clinical Update 2020. *JBMR Plus*, 5(1), e10447. <u>https://doi.org/10.1002/jbm4.10447</u>

- Misra, B., & Silverberg, S. J. (2010). Alterations in Mineral Metabolism in the Aging Male. In E. S. Orwoll, J. P. Bilezikian, & D. Vanderschueren (Eds.), Osteoporosis in Men: The Effects of Gender on Skeletal Health (Second Edition, pp. 185–192). Elsevier Academic Press.
- Mithal, A., Wahl, D. A., Bonjour, J.-P., Burckhardt, P., Dawson-Hughes, B., Eisman, J. A., El-Hajj Fuleihan, G., Josse, R. G., Lips, P., & Morales-Torres, J. (2009). Global vitamin D status and determinants of hypovitaminosis D. *Osteoporosis International*, 20(11), 1807– 1820. <u>https://doi.org/10.1007/s00198-009-0954-6</u>
- Møller, U. K., Við Streym, S., Mosekilde, L., & Rejnmark, L. (2012). Changes in bone mineral density and body composition during pregnancy and postpartum. A controlled cohort study. *Osteoporosis International*, 23(4), 1213–1223. <u>https://doi.org/10.1007/s00198-011-1654-6</u>
- Morris, H. A., Turner, A. G., & Anderson, P. H. (2012). Vitamin-D regulation of bone mineralization and remodelling during growth. *Frontiers in Bioscience-Elite*, 4(2), Article 2. <u>https://doi.org/10.2741/e409</u>
- Mundy, G. R. (2007). Osteoporosis and Inflammation. *Nutrition Reviews*, *65*(12), S147–S151. <u>https://doi.org/10.1111/j.1753-4887.2007.tb00353.x</u>
- Need, A. G., Horowitz, M., Morris, H. A., Moore, R., & Nordin, C. (2007). Seasonal Change in Osteoid Thickness and Mineralization Lag Time in Ambulant Patients. *Journal of Bone and Mineral Research*, 22(5), 757–761. <u>https://doi.org/10.1359/jbmr.070203</u>
- Nelson, D. A., & Villa, M. L. (2003). Ethnic Differences in Bone Mass and Architecture. In S. C. Agarwal & S. D. Stout (Eds.), *Bone Loss and Osteoporosis: An Anthropological Perspective* (pp. 47–62). Springer Science+Business Media.
- Norman, K., Haß, U., & Pirlich, M. (2021). Malnutrition in Older Adults—Recent Advances and Remaining Challenges. *Nutrients*, *13*(8), 2764. <u>https://doi.org/10.3390/nu13082764</u>
- O'Brien, E. C., Geraghty, A. A., Kilbane, M. T., McKenna, M. J., & McAuliffe, F. M. (2021). Bone resorption and dietary calcium in pregnancy—A window to future maternal bone health. *Osteoporosis International*, *32*(9), 1803–1814. <u>https://doi.org/10.1007/s00198-021-05891-w</u>
- Ouellet, F. (1980). *Economic and Social History of Quebec, 1760-1850: Structures and Conjonctures*. Gage Publishing Ltd.

- Parfitt, A. M. (2003). New Concepts of Bone Remodeling: A Unified Spatial and Temporal Model with Physiologic and Pathophysiologic Implications. In S. C. Agarwal & S. D. Stout (Eds.), *Bone Loss and Osteoporosis: An Anthropological Perspective* (pp. 3–18). Springer Science+Business Media.
- Parfitt, A. M. (2013). Skeletal Heterogeneity and the Purposes of Bone Remodeling: Implications for the Understanding of Osteoporosis. In R. Marcus, D. Feldman, D. Dempster, M. Luckey, & J. Cauley (Eds.), Osteoporosis (Fourth Edition, pp. 855–872). Elsevier Science & Technology.
- Pearson, D., Kaur, M., San, P., Lawson, N., Baker, P., & Hosking, D. (2004). Recovery of pregnancy mediated bone loss during lactation. *Bone*, *34*(3), 570–578. <u>https://doi.org/10.1016/j.bone.2003.11.005</u>
- Poole, A. R., Laverty, S., & Mwale, F. (2004). Endochondral bone formation and development in the axial and appendicular skeleton. In J. E. Henderson & D. Goltzman (Eds.), *The Osteoporosis Primer* (pp. 3–17). Cambridge University Press.
- Poulsen, L. W., Qvesel, D., Brixen, K., Vesterby, A., & Boldsen, J. L. (2001). Low bone mineral density in the femoral neck of medieval women: A result of multiparity? *Bone*, 28(4), 454–458. <u>https://doi.org/10.1016/S8756-3282(01)00407-0</u>
- Prentice, A., & Bates, C. J. (1993). An Appraisal of the Adequacy of Dietary Mineral Intakes in Developing Countries for Bone Growth and Development in Children. *Nutrition Research Reviews*, 6(1), 51–69. <u>https://doi.org/10.1079/NRR19930006</u>
- Prestwood, K. M., & Raisz, L. G. (2004). Consequences of alterations in bone remodeling. In J. E. Henderson & D. Goltzman (Eds.), *The Osteoporosis Primer* (pp. 199–210). Cambridge University Press.
- Priemel, M., von Domarus, C., Klatte, T. O., Kessler, S., Schlie, J., Meier, S., Proksch, N., Pastor, F., Netter, C., Streichert, T., Püschel, K., & Amling, M. (2010). Bone mineralization defects and vitamin D deficiency: Histomorphometric analysis of iliac crest bone biopsies and circulating 25-hydroxyvitamin D in 675 patients. *Journal of Bone and Mineral Research*, 25(2), 305–312. <u>https://doi.org/10.1359/jbmr.090728</u>
- Ralston, S. H. (2010). Genetics of osteoporosis. *Annals of the New York Academy of Sciences*, *1192*(1), 181–189. <u>https://doi.org/10.1111/j.1749-6632.2009.05317.x</u>
- Ralston, S. H., & Crombrugghe, B. de. (2006). Genetic regulation of bone mass and susceptibility to osteoporosis. *Genes & Development, 20*(18), 2492–2506. <u>https://doi.org/10.1101/gad.1449506</u>

- Rauch, F., & Glorieux, F. H. (2004). Osteoporosis in children. In J. E. Henderson & D. Goltzman (Eds.), *The Osteoporosis Primer* (pp. 186–196). Cambridge University Press.
- Riancho, J. A., & Brennan-Olsen, S. L. (2017). The Epigenome at the Crossroad Between Social Factors, Inflammation, and Osteoporosis Risk. *Clinical Reviews in Bone and Mineral Metabolism*, 15, 59–68. <u>https://doi.org/10.1007/s12018-017-9229-5</u>
- Riggs, B. L., Khosla, S., & Melton III, L. J. (1998). A Unitary Model for Involutional Osteoporosis: Estrogen Deficiency Causes Both Type I and Type II Osteoporosis in Postmenopausal Women and Contributes to Bone Loss in Aging Men. *Journal of Bone and Mineral Research*, 13(5), 763–773. <u>https://doi.org/10.1359/jbmr.1998.13.5.763</u>
- Ringe, J. D. (2020). Plain vitamin D or active vitamin D in the treatment of osteoporosis: Where do we stand today? *Archives of Osteoporosis*, *15*(1), 182. <u>https://doi.org/10.1007/s11657-020-00842-0</u>
- Robichaud, L. (2012). Relier Québec et Montréal par le chemin du Roy. *Cap-Aux-Diamants, 111*, 4–9.
- Russell, C. (2007). What Do Older Women and Men Want?: Gender Differences in the 'Lived Experience' of Ageing. *Current Sociology*, 55(2), 173–192. <u>https://doi.org/10.1177/0011392107073300</u>
- Scheuer, L., & Black, S. (2004). Developmental Juvenile Osteology. Elsevier Academic Press.
- Schnell, S., Friedman, S. M., Mendelson, D. A., Bingham, K. W., & Kates, S. L. (2010). The 1-Year Mortality of Patients Treated in a Hip Fracture Program for Elders. *Geriatric Orthopaedic Surgery & Rehabilitation*, 1(1), 6–14. <u>https://doi.org/10.1177/2151458510378105</u>
- Schultz, K., & Wolf, J. M. (2019). Emerging Technologies in Osteoporosis Diagnosis. *The Journal* of Hand Surgery, 44(3), 240–243. <u>https://doi.org/10.1016/j.jhsa.2018.07.006</u>
- Schultz, M. (2003). Differential Diagnoses of Intravitam and Postmortem Bone Loss at the Micro-Level. In S. C. Agarwal & S. D. Stout (Eds.), Bone Loss and Osteoporosis: An Anthropological Perspective (pp. 173–187). Springer Science+Business Media.
- Scriver, C. R. (2001). Human Genetics: Lessons from Quebec Populations. *Annual Review of Genomics and Human Genetics*, 2(1), 69–101. https://doi.org/10.1146/annurev.genom.2.1.69

- Seeman, E. (2002). Pathogenesis of bone fragility in women and men. *The Lancet, 359*(9320), 1841–1850. https://doi.org/10.1016/S0140-6736(02)08706-8
- Seeman, E., & Delmas, P. D. (2006). Bone Quality—The Material and Structural Basis of Bone Strength and Fragility. *New England Journal of Medicine*, *354*(21), 2250–2261. <u>https://doi.org/10.1056/NEJMra053077</u>
- Shapiro, S. S., & Wilk, M. B. (1965). An Analysis of Variance Test for Normality (Complete Samples). *Biometrika*, *52*(3/4), 591–611.
- Shavers, V. L. (2007). Measurement of Socioeconomic Status in Health Disparities Research. Journal of the National Medical Association, 99(9), 1013–1023.
- Sheriff, S. (1849). Final report of the Board of Health of Quebec.
- Singh, D., Sanyal, & Chattopadhyay, N. (2010). The role of estrogen in bone growth and formation: Changes at puberty. *Cell Health and Cytoskeleton*, 1. <u>https://doi.org/10.2147/CHC.S8916</u>
- Siris, E. S., Boonen, S., Mitchell, P. J., Bilezikian, J., & Silverman, S. (2012). What's in a name? What constitutes the clinical diagnosis of osteoporosis? *Osteoporosis International*, 23(8), 2093–2097. <u>https://doi.org/10.1007/s00198-012-1991-0</u>
- Stini, W. A. (2003). Bone Loss, Fracture Histories, and Body Composition Characteristics of Older Males. In S. C. Agarwal & S. D. Stout (Eds.), Bone Loss and Osteoporosis: An Anthropological Perspective (pp. 63–90). Springer Science+Business Media.
- Stodder, Ann. L. W., & Palkovich, Ann. M. (2012). Osteobiography and Bioarchaeology. In Ann.
 L. W. Stodder & Ann. M. Palkovich (Eds.), *The Bioarchaeology of Individuals* (pp. 1– 8). University Press of Florida.
- Streeter, M., & Stout, S. D. (2003). The Histomorphometry of the Subadult Rib: Age-Associated Changes in Bone Mass and the Creation of Peak Bone mass. In S. C. Agarwal & S. D. Stout (Eds.), *Bone Loss and Osteoporosis: An Anthropological Perspective* (pp. 91– 101). Springer Science+Business Media.
- Suchmacher, M., & Geller, M. (2021). *Practical Biostatistics: A Step-by-Step Approach for Evidence-Based Medicine* (Second Edition). Elsevier Academic Press.

- Szulc, P., Seeman, E., Duboeuf, F., Sornay-Rendu, E., & Delmas, P. D. (2006). Bone Fragility: Failure of Periosteal Apposition to Compensate for Increased Endocortical Resorption in Postmenopausal Women. *Journal of Bone and Mineral Research*, 21(12), 1856–1863. <u>https://doi.org/10.1359/jbmr.060904</u>
- Thapar, N., & Sanderson, I. R. (2004). Diarrhoea in children: An interface between developing and developed countries. *The Lancet*, *363*(9409), 641–653. <u>https://doi.org/10.1016/S0140-6736(04)15599-2</u>
- Uday, S., & Högler, W. (2019). Spot the silent sufferers: A call for clinical diagnostic criteria for solar and nutritional osteomalacia. *The Journal of Steroid Biochemistry and Molecular Biology*, *188*, 141–146. <u>https://doi.org/10.1016/j.jsbmb.2019.01.004</u>
- van Gerven, D. P., Armelagos, G. J., & Bartley, M. H. (1969). Roentgenographic and direct measurement of femoral cortical involution in a prehistoric Mississippian population. *American Journal of Physical Anthropology*, *31*(1), 23–38. <u>https://doi.org/10.1002/ajpa.1330310105</u>
- van Manen, M. (1997). *Researching Lived Experience: Human Science for an Action Sensitive Pedagogy* (Second Edition). Routledge, Taylor & Francis Group.
- van Spelde, A.-M., Schroeder, H., Kjellström, A., & Lidén, K. (2021). Approaches to osteoporosis in paleopathology: How did methodology shape bone loss research? *International Journal of Paleopathology*, 33, 245–257. <u>https://doi.org/10.1016/j.ijpp.2021.05.001</u>
- Vigeant, J., Ribot, I., & Hélie, J.-F. (2017). Dietary habits in New France during the 17th and 18th centuries. *American Journal of Physical Anthropology*, *162*(3), 462–475. <u>https://doi.org/10.1002/ajpa.23129</u>
- Waldron, T. (1987). The relative survival of the human skeleton: Implications for paleopathology. In A. Boddington, A. N. Garland, & R. C. Janaway (Eds.), *Death, decay, and reconstruction: Approaches to archaeology and forensic science* (pp. 55–64). Manchester University Press.
- Wallace, I. J., Demes, B., & Judex, S. (2017). Ontogenetic and Genetic Influences on Bone's Responsiveness to Mechanical Signals. In C. J. Percival & J. T. Richtsmeier (Eds.), *Building Bones: Bone Formation and Development in Anthropology* (pp. 233–253). Cambridge University Press.

- Weiss, C. O. (2011). Frailty and Chronic Diseases in Older Adults. *Clinics in Geriatric Medicine*, 27(1), 39–52. <u>https://doi.org/10.1016/j.cger.2010.08.003</u>
- Welsh, H., Nelson, A. J., van der Merwe, A. E., de Boer, H. H., & Brickley, M. B. (2020). An Investigation of Micro-CT Analysis of Bone as a New Diagnostic Method for Paleopathological Cases of Osteomalacia. *International Journal of Paleopathology*, 31, 23–33. <u>https://doi.org/10.1016/j.ijpp.2020.08.004</u>
- WHO Scientific Group. (2003). *Prevention and Management of Osteoporosis: Report of a WHO Scientific Group* (WHO Technical Report Series). World Health Organization.
- Wiren, K. (2005). Androgens and bone growth: It's location, location, location. *Current Opinion* in Pharmacology, 5(6), 626–632. <u>https://doi.org/10.1016/j.coph.2005.06.003</u>
- Wissler, A., Blevins, K. E., & Buikstra, J. E. (2022). Missing data in bioarchaeology II: A test of ordinal and continuous data imputation. *American Journal of Biological Anthropology*, *179*(3), 349–364. <u>https://doi.org/10.1002/ajpa.24614</u>
- Wood, J. W., Milner, G. R., Harpending, H. C., Weiss, K. M., Cohen, M. N., Eisenberg, L. E., Hutchinson, D. L., Jankauskas, R., Cesnys, G., Katzenberg, M. A., Lukacs, J. R., McGrath, J. W., Roth, E. A., Ubelaker, D. H., & Wilkinson, R. G. (1992). The Osteological Paradox: Problems of Inferring Prehistoric Health from Skeletal Samples [and Comments and Reply]. *Current Anthropology*, *33*(4), 343–370. <u>https://doi.org/10.1086/204084</u>
- Wright, N. C., Looker, A. C., Saag, K. G., Curtis, J. R., Delzell, E. S., Randall, S., & Dawson-Hughes, B. (2014). The Recent Prevalence of Osteoporosis and Low Bone Mass in the United States Based on Bone Mineral Density at the Femoral Neck or Lumbar Spine: RECENT US PREVALENCE OF OSTEOPOROSIS AND LOW BONE MASS. *Journal of Bone* and Mineral Research, 29(11), 2520–2526. <u>https://doi.org/10.1002/jbmr.2269</u>
- Yuan, Y., Zhang, L., Tong, X., Zhang, M., Zhao, Y., Guo, J., Lei, L., Chen, X., Tickner, J., Xu, J., & Zou, J. (2017). Mechanical Stress Regulates Bone Metabolism Through MicroRNAs: MECHANICAL STRESS REGULATES BONE GROWTH VIA MiRNAs. *Journal of Cellular Physiology*, 232(6), 1239–1245. <u>https://doi.org/10.1002/jcp.25688</u>
- Zaki, M. E., Hussien, F. H., & El Banna, R. A. E.-S. (2009). Osteoporosis among ancient Egyptians. International Journal of Osteoarchaeology, 19(1), 78–89. <u>https://doi.org/10.1002/oa.978</u>

- Zebaze, R., & Seeman, E. (2010). Age-related Changes in Bone Remodeling and Microarchitecture. In E. S. Orwoll, J. P. Bilezikian, & D. Vanderschueren (Eds.), Osteoporosis in Men: The Effects of Gender on Skeletal Health (Second Edition, pp. 167–178). Elsevier Academic Press.
- Zuckerman, M., & Martin, D. (2016). Introduction: The development of biocultural perspectives in anthropology. In M. Zuckerman & D. Martin (Eds.), *New Direction in Biocultural Anthropology* (pp. 7–28). John Wiley & Sons, Inc.

Appendices
Appendix A: Skeletal Sample

Age, Sex, and Cortical Area Index (CAI_{ROI}) Data

Individual	Sex	Age	Age	CAI _{ROI}
		Estimate	Category	
Pointe-aux-				
Trembles			1	
7A11-S42	F	14-20	15-29	27.813
7A9-S13	F	15-24	15-29	61.132
7A9-S38	М	14-19	15-29	67.181
7A9-S45	F	18-25	15-29	26.518
7A2-S20	F	20-29	15-29	29.382
7A9-S9	F	25-29	15-29	53.959
7A9-S27	F	23-35	15-29	57.575
7A11-S60	F	>25	15-29	61.848
7A11-S61	F	15-24	15-29	62.069
7A2-S5	М	30-34	30-49	52.715
7A2(9)-S23	F	30-34	30-49	61.808
7A2-S2	F	23-35	30-49	69.045
7A2-S11	F	41-50	30-49	32.956
7A9-S16	М	>24	30-49	47.406
7A9-S31	М	50-63	50+	45.289
7A11-S19	М	50-77	50+	43.313
7A9-S15	М	66-80	50+	44.456
7A9-S46	F	60-87	50+	49.569
Sainte-Marie				
2F4	Μ	14-19	15-29	49.570
2E10	F	14-15	15-29	57.231
2E13	F	16-20	15-29	73.856
2B7	Μ	20-30	15-29	37.543
2B15	Μ	20-30	15-29	37.637
2G4	Μ	20-24	15-29	43.449
2B8	Μ	25-29	15-29	50.846
2F	Μ	20-24	15-29	51.892
2B14	F	20-29	15-29	52.520
2 E8	М	20-24	15-29	53.996
2G6	F	20-39	30-49	40.266
1G5	F	35-39	30-49	55.672
2E14	F	30-39	30-49	57.066
2B16	F	>25	30-49	41.861
2A14	F	40-50	30-49	48.914
2B9	F	50-60	50+	31.012

Table A1. Age, Sex, and Cortical Area Index (CAI_{ROI}) data

2B11	F	45-59	50+	49.130
2G2	М	>55	50+	54.598
2B3	F	60+	50+	42.900
2E11	F	60+	50+	49.796
2G12	F	60+	50+	58.929
2E12	F	80+	50+	42.154
2F1	М	80+	50+	42.187
2B10	М	80+	50+	46.631
2E9	F	80+	50+	57.808

Appendix B: Skeletal Recording Forms

B1: Age and Sex Recording Form

B2: Metacarpal Radiogrammetry Recording Form

B1: Age and Sex Recording Form

Site:	Date:
Individual:	

<u>Sex</u>

Pelvis:	Skull:
Ventral Arc (1-3)	Nuchal Crest (1-5)
Subpubic Concavity (1-3)	Mastoid Process (1-5)
Ischiopubic Ramus Ridge (1-3)	Supraorbital Margin (1-5)
Greater Sciatic Notch (1-5)	Glabella (1-5)
Preauricular Sulcus (0-4)	Mental Eminence (1-5)
Estimated Sex (Pelvis):	Estimated Sex (Skull):

<u>Age</u>

Epiphyseal Fusion

Pubic Symphysis (1-6)

Auricular Surface (1-8) _____

Transition Analysis:

-Iliac Exostoses _____

-Dorsal Margin

Estimated Age: _____

Notes/Comments:

B2: Metacarpal Radiogrammetry Recording Form

Site:	Date:
Individual:	
Left: Right:	
Complete:	
<25% 25%-50% 50%-75% >75%	
Cortical bone area (CB):	

Total bone area (TA):

ROI cortical area index (CA_{ROI}):

Appendix C: Comparative Analysis

Reference	Location of Study	Time Period	N	Methods	Author Conclusions
Agarwal 2012	England (Rural)	11th-16th century	f=30, m=24	Trabecular microstructure (L4)	No significant differences in bone amount between sexes, consistent age- related bone loss
	England (Urban)	11th-14th century	f=41, m=32	_	Significant sex and age differences: older females (50+) have significantly less bone than older males
Agarwal and Grynpas 2009	England	11th-16th century	f=31, m=27	Bone mineral density, digital x-ray absorptiometry	Males show steady age-related bone loss, females lose BMD at an earlier age
Beauchesne and Agarwal 2014	Velia, Italy	1st-2nd century	f=32, m=39	Metacarpal radiogrammetry	Age-related bone loss, no statistically significant sex differences in cortical index for each age group
Beauchesne and Agarwal 2017	Velia, Italy	1st-2nd century	f=30, m=40	Rib cross-sectional area	Follows clinical age and sex trends
			f=26, m=26	Rib histomorphometry	Bone formation rate decreases significantly between all age groups, no significant differences between sexes
			f=19, m=28	Trabecular microstructure (L4)	Trabecular spacing and number decrease with age, slight sex differences found.
Curate et al. 2013	Portugal	19th-20th century	f=98, m=98	Bone mineral density (proximal femur), digital x-ray absorptiometry	Significant sex and age differences, older females have significantly less bone than older males.

Dewey et al. 1969	Sudan	350 B.C A.D. 350 A.D. 350- A.D. 550	f=29, m=17 f=63, m=42	Cortical thickness, femur	Males follow expected age-related bone loss trends, cortical thickness in females is lowest from age 32-41 and increases by age 42-50. Follows clinical age and sex trends
		A.D. 550- A.D. 1300	f=28, m=24		Male cortical thickness increases with age, female cortical thickness fluctuates with age.
Ekenman et al. 1995	Sweden	14th-15th century	f=156, m=187	Bone mineral density, cortical thickness	No significant differences in bone amount between sexes, no age-related bone loss
Ericksen 1976	Southwestern United States	A.D. 1130	n=142	Cortical thickness, femut and humerus	Adheres to clinical age and sex trends
	Midwestern United States	A.D. 1550- A.D. 1845	n=134		
	Alaskan Coast	18th-20th century	n=123		
Glencross and Agarwal 2011	Turkey	7400 B.C 6000 B.C.	f=27, m=22	Metacarpal radiogrammetry	Adheres to clinical age and sex trends
Holck 2007	Norway	5000 B.C A.D. 800	f=13, m=23	Bone mineral density, femur	No sex differences, age-related bone loss
		A.D. 800- A.D. 1050	f=15, m=23		
		A.D. 1050- A.D. 1536	f=54, m=57		

M.A. Thesis- R. Christenson, McMaster University - Anthropology

Poulsen et al. 2001	Denmark	A.D. 1000- A.D. 1250	f=20, m=29	Bone mineral density, femoral neck	No age-related bone loss in males, increase in bone mineral density with age in females.
Van Gerven et al. 1969	Midwestern United States	A.D. 1540- A.D. 1700	f=23, m=20	X-ray measurements of femoral cortical thickness, periosteal diameter	Age-related bone loss in males and females, females lost significantly more bone than males
Zaki et al. 2009	Egypt	2687 B.C 2191 B.C.	f=31, m=43	Bone mineral density, various skeletal sites	Significant age-related bone loss in males and females, females lose bone earlier than males.

Appendix D: Metacarpal Radiogrammetry

D1: Pointe-aux-Trembles Data and Radiographs

D2: Sainte-Marie Data and Radiographs

D1: Pointe-aux-Trembles

Individual	Cortical	Total	CAI _{ROI}	Radiograph
	Bone Area	Bone		
		Area		
7A11-S42	43.758	157.332	27.813	D1.2
7A9-S13	95.748	156.625	61.132	D1.3
7A9-S38	108.642	161.715	67.181	D1.3
7A9-S45	43.133	162.653	26.518	D1.3
7A2-S20	40.659	138.382	29.382	D1.3
7A9-S9	83.771	155.248	53.959	D1.3
7A9-S27	83.998	145.839	57.575	D1.1
7A11-S60	80.983	130.938	61.848	D1.1
7A11-S61	99.943	161.020	62.069	D1.2
7A2-S5	88.969	168.774	52.715	D1.1
7A2(9)-S23	87.716	141.918	61.808	D1.2
7A2-S2	97.338	140.976	69.045	D1.2
7A2-S11	43.606	132.316	32.956	D1.1
7A9-S16	85.463	180.279	47.406	D1.4
7A9-S31	76.511	168.938	45.289	D1.2
7A11-S19	73.219	169.046	43.313	D1.4
7A9-S15	84.776	190.697	44.456	D1.1
7A9-S46	70.377	141.979	49.569	D1.4

Table D1.1. Metacarpal Bone Measurements for the Pointe-aux-Trembles sample



Figure D1.1. Metacarpal radiographs of individual 7A2-S5, 7A9-S27, 7A9-S15, 7A2-S11, and 7A11-S60



Figure D1.2. Metacarpal radiographs of individual 7A2-S2, 7A2(9)-S23, 7A9-S31, 7A11-S42, and 7A11-S61



Figure D1.3. Metacarpal radiographs of individual 7A2-S20, 7A9-S13, 7A9-S9, 7A9-S38, and 7A9-S45



Figure D1.4. Metacarpal radiographs of individuals 7A9-S46, 7A9-S16, and 7A11-S19

D2: Sainte-Marie

Individual	Cortical Bone	Total		Radiograph
	Area	Bone		
		Area		
2F4	71.958	145.165	49.570	D2.1
2E10	84.093	146.936	57.231	D2.1
2E13	94.161	127.493	73.856	D2.5
2B7	78.545	209.213	37.543	D2.4
2B15	76.135	202.286	37.637	D2.6
2G4	69.308	159.514	43.449	D2.6
2B8	92.166	181.266	50.846	D2.2
2F	90.984	175.333	51.892	D2.2
2B14	76.553	145.761	52.520	D2.4
2E8	103.612	191.89	53.996	D2.2
2G6	70.484	175.048	40.266	D2.5
1G5	76.076	136.65	55.672	D2.6
2E14	85.494	149.816	57.066	D2.5
2B16	57.495	137.349	41.861	D2.4
2A14	63.467	129.751	48.914	D2.1
2B9	39.771	128.244	31.012	D2.3
2B11	77.915	158.588	49.130	D2.3
2G2	112.897	206.778	54.598	D2.4
2B3	63.033	146.931	42.900	D2.5
2E11	73.856	148.316	49.796	D2.4
2G12	80.655	145.351	58.929	D2.3
2E12	55.845	132.477	42.154	D2.2
2F1	70.671	167.518	42.187	D2.3
2B10	101.567	217.808	46.631	D2.2
2E9	80.907	139.958	57.808	D2.5

Table D2.1. Metacarpal Bone Measurements for the Sainte-Maries sample



Figure D2.1. Metacarpal radiographs of individual 2A14, 2E10, and 2F4



Figure D2.2. Metacarpal radiographs of individual 2E8, 2E12, 2B10, 2B8, and 2F



Figure D2.3. Metacarpal radiographs of individuals 2B9, 2G12, 2B11, and 2F1



Figure D2.4. Metacarpal radiographs of individual 2B16, 2B14, 2B7, 2E11, and 2G2



Figure D2.5. Metacarpal radiographs of individual 2E13, 2E14, 2G6, 2E9, and 2B3



Figure D2.6. Metacarpal radiographs of individual 1G5, 2B15, and 2G4

Appendix E: Statistical Data

E1: Shapiro-Wilk Normality Tests

E2: T-Test

E3: Descriptive Statistics

E4: ANOVA

E1: Shapiro-Wilk Normality Tests

	Sex	Shapiro-Wilk			
		Statistic <i>df</i> Sig.			
CAI _{ROI}	Male	.933	16	.271	
	Female	.951 27 .230			

Table E1.1. Shapiro-Wilk normality test for each sex category

 Table E1.2.
 Shaprio-Wilk normality test for each age category

	Age	Shapiro-Wilk			
		Statistic	df	Sig.	
CAI _{ROI}	15-29	.949	19	.374	
	30-49	.993	10	.999	
	50+	.944 14 .468			

E2: Student's T-Test

Table E2.1.	T-test for	comparing	sex and	cortical	area in	dex (
	1 1051 101	comparing	JCA una	conticui	ui cu ili	acr (

Age Category	<i>t</i> -value	Degrees of Freedom	<i>p</i> -value
15-29	-0.352	17	0.729
30-49	-0.098	8	0.924
50+	-0.0393	12	0.702
Total Sample	-0.671	40.992	0.506

E3: Descriptive Statistics

 Table E3.1. Descriptive statistics for cortical area index (CAI_{ROI}) by age category

	N	Range	Minimum	Maximum	Mean	Standard
						Deviation
15-29	19	47.338	26.518	73.856	50.317	13.425
30-49	10	36.089	32.956	69.045	50.771	10.776
50+	14	27.917	31.012	58.929	46.984	7.222
Total	43	47.338	26.518	73.856	49.337	11.002
Sample						

E4: ANOVA

Multiple Comparisons						
Age	Age	Mean	Std.	Sig.	95% Confidence Interval	
Group 1	Group	Difference	Error		Lower	Upper
	2				Bound	Bound
15-29	30-49	454216	4.353622	.994	-11.05058	10.14215
	50+	3.332970	3.925045	.675	-6.22027	12.88621
30-49	15-29	.454216	4.353622	.994	-10.14215	11.05058
	50+	3.787186	4.613920	.692	-7.44272	15.01710
50+	15-29	-3.332970	3.925045	.675	-12.88621	6.22027
	30-49	-3.787186	4.613920	.692	-15.01710	7.44272

Table E4.1. ANOVA results for the comparison of cortical area index (CAI_{ROI}) and each age group

Appendix F: Micro-CT Data

- F1: 7A2-S2 Scores and 5 Scan Slices
- F2: 7A2-S11 Scores and 5 Scan Slices
- F3: 7A2-S2 Scores and 5 Scan Slices
- F4: 7A9-S38 Scores and 5 Scan Slices
- F5: 7A9-S46 Scores and 5 Scan Slices
- F6: 7A11-S19 Scores and 5 Scan Slices
- F7: 7A11-S42 Scores and 5 Scan Slices
 - F8: 2B9 Scores and 5 Scan Slices
 - F9: 2E13 Scores and 5 Scan Slices
 - F10: 2G4 Scores and 5 Scan Slices

F1: 7A2-S2 Scores and 5 Scans

Table F1.1. Slice scores for individual 7A2-S2

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
300	1	0	1.0
550	1	0	1.0
800	1	0	1.0
1050	1	0.5	1.5
1300	2	0	2.0
	6	0.5	6.5



Figure F1.1. Micro-CT of individual 7A2-S2, slice 300



Figure F1.2. Micro-CT of individual 7A2-S2, slice 550



Figure F1.3. Micro-CT of individual 7A2-S2, slice 800



Figure F1.4. Micro-CT of individual 7A2-S2, slice 1050



Figure F1.1. Micro-CT of individual 7A2-S2, slice 1300

F2: 7A2-S11 Scores and 5 Scan Slices

Table F2.1. Slice scores for individual 7A2-S11

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
500	1	0.5	1.5
750	1	0.5	1.5
1000	1	0.5	1.5
1250	3	0.5	3.5
1500	2	0.5	2.5
	8	2.5	10.5



Figure F2.1. Micro-CT of individual 7A2-S11, slice 500



Figure F2.2. Micro-CT of individual 7A2-S11, slice 750



Figure F2.3. Micro-CT of individual 7A2-S11, slice 1000


Figure F2.4. Micro-CT of individual 7A2-S11, slice 1250



Figure F2.5. Micro-CT of individual 7A2-S11, slice 1500

F3: 7A2-S20 Scores and 5 Scan Slices

Table F3.1. Slice scores for individual 7A2-S20

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
350	0	0	0.0
600	1	0	1.0
850	1	0.5	1.5
1100	1	0.5	1.5
1350	1	0.5	1.5
	4	1.5	5.5



Figure F3.1. Micro-CT of individual 7A2-S20, slice 350



Figure F3.2. Micro-CT of individual 7A2-S20, slice 600



Figure F3.3. Micro-CT of individual 7A2-S20, slice 850



Figure F3.4. Micro-CT of individual 7A2-S20, slice 1100



Figure F3.5. Micro-CT of individual 7A2-S20, slice 1350

F4: 7A9-S38 Scores and 5 Scan Slices

Table F4.1. Slice scores for individual 7A9-S38

Slice Number	DCL Score	Incomplete Mineralization(+0.5)	Total Score
550	0	0	0.0
800	0	0	0.0
1050	0	0	0.0
1300	0	0	0.0
1550	0	0	0.0
	0	0	0.0



Figure F4.1. Micro-CT of individual 7A9-S38, slice 550



Figure F4.2. Micro-CT of individual 7A9-S38, slice 800



Figure F4.3. Micro-CT of individual 7A9-S38, slice 1050



Figure F4.4. Micro-CT of individual 7A9-S38, slice 1300



Figure F4.5. Micro-CT of individual 7A9-S38, slice 1550

F5: 7A9-S46 Scores and 5 Scan Slices

Table F5.1. Slice scores for individual 7A9-S46

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
450	2	0.5	2.5
700	1	0	1.0
950	2	0	2.0
1200	1	0	1.0
1450	0	0	0.0
	6	0.5	6.5



Figure F5.1. Micro-CT of individual 7A9-S46, slice 450



Figure F5.2. Micro-CT of individual 7A9-S46, slice 700



Figure F5.3. Micro-CT of individual 7A9-S46, slice 950



Figure F5.4. Micro-CT of individual 7A9-S46, slice 1200



Figure F5.5. Micro-CT of individual 7A9-S46, slice 1450

F6: 7A11-S19 Scores and 5 Scan Slices

Table F6.1. Slice scores for individual 7A11-S19

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
650	2	0.5	2.5
900	3	0.5	3.5
1150	1	0.5	1.5
1400	2	0.5	2.5
1650	2	0.5	2.5
	10	2.5	12.5



Figure F6.1. Micro-CT of individual 7A11-S19, slice 650



Figure F6.2. Micro-CT of individual 7A11-S19, slice 900



Figure F6.3. Micro-CT of individual 7A11-S19, slice 1150



Figure F6.4. Micro-CT of individual 7A11-S19, slice 1400



Figure F6.5. Micro-CT of individual 7A11-S19, slice 1650

F7: 7A11-S42 Scores and 5 Scan Slices

Table F7.1. Slice scores for individual 7A11-S42

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
550	1	0	1.0
800	1	0	1.0
1050	1	0.5	1.5
1300	1	0	1.0
1550	0	0	0.0
	4	0.5	4.5



Figure F7.1. Micro-CT of individual 7A11-S42, slice 550

M.A. Thesis- R. Christenson, McMaster University - Anthropology



Figure F7.2. Micro-CT of individual 7A11-S42, slice 800



Figure F7.3. Micro-CT of individual 7A11-S42, slice 1050



Figure F7.4. Micro-CT of individual 7A11-S42, slice 1300



Figure F7.5. Micro-CT of individual 7A11-S42, slice 1550

F8: 2B9 Scores and 5 Scan Slices

Table F8.1. Slice scores for individual 2B9

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
300	0	0	0.0
550	0	0	0.0
800	0	0	0.0
1050	0	0	0.0
1300	1	0	1.0
	1	0	1.0



Figure F8.1. Micro-CT of individual 2B9, slice 300



Figure F8.2. Micro-CT of individual 2B9, slice 550



Figure F8.3. Micro-CT of individual 2B9, slice 800



Figure F8.4. Micro-CT of individual 2B9, slice 1050



Figure F8.5. Micro-CT of individual 2B9, slice 1300

F9: 2E13 Scores and 5 Scan Slices

Table F9.1. Slice scores for individual 2E13

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
800	2	0.5	2.5
1050	2	0.5	2.5
1300	3	0.5	3.5
1550	2	0.5	2.5
1800	3	0.5	3.5
	12	2.5	14.5



Figure F9.1. Micro-CT of individual 2E13, slice 800



Figure F9.2. Micro-CT of individual 2E13, slice 1050



Figure F9.3. Micro-CT of individual 2E13, slice 1300



Figure F9.4. Micro-CT of individual 2E13, slice 1550


Figure F9.5. Micro-CT of individual 2E13, slice 1800

F10: 2G4 Scores and 5 Scan Slices

Table F10.1. Slice scores for individual 2G4

Slice Number	DCL Score	Incomplete	Total Score
		Mineralization(+0.5)	
450	0	0	0.0
700	1	0	1.0
950	1	0	1.0
1200	0	0	0.0
1450	0	0	0.0
	2	0	2.0



Figure F10.1. Micro-CT of individual 2G4, slice 450



Figure F10.2. Micro-CT of individual 2G4, slice 700



Figure F10.3. Micro-CT of individual 2G4, slice 950



Figure F10.4. Micro-CT of individual 2G4, slice 1200



Figure F10.5. Micro-CT of individual 2G4, slice 1450